

A Service of

ZBU

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Kubik, Zaneta

Working Paper The challenges of rural youth employment in Africa: A literature review

ZEF Working Paper Series, No. 212

Provided in Cooperation with: Zentrum für Entwicklungsforschung / Center for Development Research (ZEF), University of Bonn

Suggested Citation: Kubik, Zaneta (2022) : The challenges of rural youth employment in Africa: A literature review, ZEF Working Paper Series, No. 212, University of Bonn, Center for Development Research (ZEF), Bonn

This Version is available at: https://hdl.handle.net/10419/264365

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Working Paper 212

ΖΑΝΕΤΑ Κυβικ

The challenges of rural youth employment in Africa: a literature review





ZEF Working Paper Series, ISSN 1864-6638 Center for Development Research, University of Bonn Editors: Christian Borgemeister, Joachim von Braun, Manfred Denich, Till Stellmacher and Eva Youkhana

Author's address

Zaneta Kubik Center for Development Research (ZEF), University of Bonn Genscherallee 3 53113 Bonn, Germany zkubik@uni-bonn.de

The challenges of rural youth employment in Africa: a literature review

Zaneta Kubik

Abstract

This paper provides a literature review on youth employment challenges in rural Africa. The rapidly changing demographic structure across the continent is expected to have important economic and social consequences, especially for employment. So far, despite sustained economic growth, African countries have not been successful in absorbing the fast-growing labor force, especially in the context of labor markets characterized by high levels of informality, underemployment and working poverty. As a consequence of life-cycle effects, relative to the measure of accumulated life experience; and generational effects, relative to the measure of the conditions prevailing during an individual's formative years, young people are exposed to several constraints in the labor market, including access to resources such as skills and innovation, finance and land. These challenges call for a comprehensive policy framework with complementary supply-side and demand-side interventions. Interventions that target girls and women can have particularly strong effects on their labor market outcomes.

Food systems are increasingly recognized as potential catalyst for employment creation, given their future prospects and labor-intensive nature. Farming and self-employment in the agri-food sector are the dominant categories of youth employment in rural areas of Africa, and the latter is growing especially fast, even though it remains much lower than farming in absolute terms. Despite common perceptions, food system jobs play a significant role in youth's aspirations. Close to 25% of young Africans want to work in the food and agriculture sector, and the share is higher in some countries, close to 40% in Kenya, Liberia, Malawi and Tanzania. Accordingly, the average age of African farmers is not rising – it is also much lower than previously claimed, at 34 years of age and not 60. However, youth aspirations remain conditional on several factors that can make food system jobs attractive, including technology, investment, market opportunities, and decent earnings. Policies should prioritize interventions that will raise labor productivity in food system, along with the broader labor-market interventions.

Keywords: youth, labor force, employment, food systems, agriculture

JEL codes: J13, J21, J22, J43, J48

Acknowledgments

The author wants to thank Joachim von Braun and Heike Baumüller for helpful comments, and Laura Newton for editorial assistance.

This study was developed in the context of the Program of Accompanying Research on Agricultural Innovation (PARI), supported by the Federal German Ministry for Economic Cooperation and Development (BMZ).

Table of contents

III IV E				
١V				
E				
5				
7				
10				
10				
11				
11				
12				
13				
14				
14				
4.2Food systems in youth aspirations14				
4.3Status of youth employment in food systems17				
18				
19				
21				
21				
22				
22				
24				
34				
36				

1 Introduction

Demographic pressure and the youth bulge have spurred much debate in Africa. With three quarters of population below the age of 35, Africa has the youngest population in the world. This is also the only region in the world where the youth population is projected to continue growing and will more than double by 2055 (UNDESA, 2019). Depending on the context in which the youth bulge occurs, and the opportunities and challenges that arise for youth's productive employment and personal development, the youth bulge could turn into a great opportunity, contributing to economic growth and structural transformation; or a great challenge, with implications for economic and social stability across the continent. In addition, the current conjuncture of climate, health and economic crises has cast a spotlight on the future prospects of young people and the growing concerns over intergenerational sustainability (HLPE, 2021).

While the potential economic and social consequences of the expected changes in the demographic structure of the African countries are multifaceted, the implications in terms of employment and local labor markets are particularly important. In Africa, the youth labor force almost doubled between 1990 and 2020, from 61.8 million to 115.8 million; and it is projected to have further grown by nearly 30 million and reached 144 million by 2030 (ILO, 2020). Considering the strenuous conditions in the local labor markets, characterized by high levels of informality, underemployment and working poverty, the projected influx of new entrants will put additional pressure on the already insufficient number of decent jobs. So far, despite sustained economic growth, especially over the last five years, labor markets in Africa have not been able to absorb the fast-growing labor force. The employment elasticity of growth has been low in Africa, with an estimated 0.41% increase in employment corresponding to a 1% increase in GDP (AfDB, 2019). Compared to other regions of the world, labor productivity remains low in Africa, which has also translated into declining real wages (ILO, 2020).

Compared to the adult population, youth is exposed to particular challenges in the labor market. At 63% in 2020, the share of young workers living in poverty is higher compared to adult workers, at 50%; and even though working poverty of the youth has been falling over the last two decades in Africa, it remains much above the levels observed in other regions of the world. Moreover, one in five of all young Africans have neither a job nor are they participating in education or training; and the vast majority of those not in employment, education or training (NEET) are young women (ILO, 2020). The particular youth challenges in the labor market are considered to stem from both life-cycle effects, i.e. relative to the measure of accumulated life experience; and generational effects, i.e. relative to the measure of the conditions prevailing during an individual's formative years.

In consequence, young people in Africa are exposed to specific barriers to entry into productive employment. These entry barriers are twofold: first, skills and qualifications, especially evidenced by skill mismatch; second, access to resources, mainly land and financial resources. For young people living in rural areas, these constraints are further exacerbated by the existing structural deficiencies of local economies and inadequacies in infrastructure and public service provision; various market imperfections tend to be amplified in remote and less connected areas. In line with the observed duality of labor markets in developing countries, rural labor markets also differ from the urban ones in that the majority of employment is concentrated in agriculture and food systems, and that informal and self-employment are prevailing forms of employment, while the formal wage-employment options are rare.

The challenge of youth employment has attracted a lot of attention among policy-makers and researchers. Creating productive employment and enabling environment for youth in Africa has come to the fore of the policy agendas in most countries. The Sustainable Development Goal (SDG) 8.6 specifically focuses on the promotion of youth employment, education and training. The African Youth Charter, adopted in 2006, was among the first instruments developed in Africa to provide a strategic framework for youth employment. The African Union (AU) Agenda 2063 also includes goals relative to youth employment. The year 2017 was nominated by the AU as the year of youth, with the objective

of harnessing the demographic dividend though investments in youth. The political commitment to the youth employment challenge is also reflected, directly or indirectly, in other strategic documents, either at the continental or country level (ILO, 2020). The research, too, has increasingly focused on youth, providing a growing evidence on various aspects of youth employment in Africa, but also other regions of the world.

This paper presents a literature review on the youth employment challenge in rural areas of Africa with the objective of addressing the question of how to productively engage rural youth. It proposes a conceptual framework to analyze the various aspects of youth employment, including determinants of labor market outcomes of youth and constraints which arise as a consequence of life-cycle and generational effects and which influence the spectrum of opportunities young people can choose from, especially when entering the labor market. The conceptual framework guides the literature review along selected thematic lines. In line with the focus of much of the available literature, emphasis is placed on employment of youth in food systems, considering its current and future prospects to create decent jobs in rural areas of Africa. However, acknowledging the increasingly complex livelihood strategies of rural youth, with evidence on pluri-locality and pluri-activity, the analysis also looks beyond food systems and rural areas and offers insights on mobility across space and between sectors of employment. Finally, the paper provides policy considerations, including the trade-offs and time-consistency aspects of required investments.

The remainder of this paper proceeds as follows. Section 2 presents the conceptual framework and method of the analysis; section 3 analyzes the macro-determinants of the youth labor markets in Africa; section 4 focuses on youth employment in food systems; section 5 discusses main constraints that young people face when entering labor markets. Finally, section 6 provides selected policy considerations and the last section concludes the review.

2 Conceptual framework and method

Youth can be defined as a period of transition from childhood and adolescence to adulthood, or from school to work. This is a particular period in an individual's lifetime, marked by the evolution of dependency relationships (Locke & te Lintelo, 2012; MacDonald, 2011) and transition to economic independence (Ansell, 2004). As such, youth is considered as a formative stage of life; the education and employment-related experiences accumulated at this stage affect the lifelong economic prospects and welfare of an individual (Filmer & Fox, 2014). For analytical purposes, youth, as well as childhood and adulthood as the remaining stages of life, are usually defined by chronological age (HLPE, 2021). The UN system defines youth as ages 15-24, and the AU as ages 15-35. However, the official definitions vary widely in national policies (Filmer & Fox, 2014; Arulingam et al., 2019). Moreover, even within the same framework, definitions of youth often overlap with definitions of childhood. In this context, some argue that childhood and youth should be considered in relative rather than absolute terms, and defined by their position in inter-generational relations and across the life cycle (Huijsmans, 2016; HLPE, 2021).

The conceptual framework in Figure 1 below is centered around the concept of transition between stages of life, i.e. childhood and adulthood, and, in more analytical terms, transition from school to work – the principal focus of this literature review. Conventional indicators of these transitions include completion of education, entry into employment, achievement of economic independence and family formation (Durham, 2017). It is important to acknowledge that although the transitions, as presented above, appear as linear movement between consecutive stages of life, their trajectories are often very diverse, complex and non-linear (Locke & te Lintelo, 2012).

The trajectories of youth transitions depend on two types of effects: first, life-cycle effects, i.e. relative to the measure of accumulated life experience; and second, generational effects, i.e. relative to the measure of the conditions prevailing during an individual's formative years (Resnick & Thurlow, 2015; Glover & Sumberg, 2020). Life-cycle effects refer to factors that are intrinsic to youth as a stage of life, including socio-economic characteristics, such as lack of work experience or control over assets. Here, these effects are conceptualized in terms of limited access to resources, including knowledge and skills, land, and finance. On the other hand, the generational effects, which are specific to the given period of time do not concern youth exclusively; nevertheless, the intensity of their impacts might be particularly high for youth in confluence with the life-cycle effects. These effects are considered here in terms of the macro-determinants of the youth labor markets, including cohort size, macroeconomic context, and labor market trends, including entrepreneurship and mechanization and automation of work.

In addition, there exist complex and dynamic relationships between different stages of life and also, different generations. In this respect, youth transitions are the outcome not only of individual choices, but also aspirations, agency, and intergenerational power dynamics. Recent theoretical and empirical works show that occupational, educational, and income aspirations strongly determine individual efforts and decisions in terms of schooling (Ashraf, 2016; Favara, 2017; Tanguy et al., 2014), uptake of productivity-enhancing technologies and credit services (Bernard, Dercon & Taffesse, 2012; Mausch et al., 2018; Mekonnen & Gerber, 2016) and entrepreneurship (Langevang et al., 2012; Mgumia, 2017; Afutu-Kotey et al., 2017). Aspirations have often been conceptualized in the literature as expectations rooted in the reality (Leavy & Smith, 2010; Daum, 2018), or "opportunity spaces" (Sumberg et al., 2012), which are formed at the confluence of geographical, political, economic and cultural factors (Bossenbroek, van der Ploeg & Zwarteveen, 2015).

How youth aspirations meet the reality is, along those contextual factors, also a question of agency and youth bargaining power within the household. In case of labor market outcomes, the labor allocation decision has been modelled in terms of utility maximization either at individual level, as in the Todaro model (Todaro, 1969), or at the household level, as in the Becker model (Becker, 1965). It

is therefore not clear to which extent livelihood choices of youths reflect their own aspirations as opposed to those of their families (Bezu & Holden, 2014).



Source: Author's own elaboration.

The employment opportunities of the rural youth, especially in the context of rural Africa, can be divided into those within and outside of the food system. Food systems can be defined as "all the elements (environment, people, inputs, processes, infrastructures, institutions, etc.) and activities that relate to the production, processing, distribution, preparation and consumption of food, and the output of these activities, including socio-economic and environmental outcomes" (HLPE, 2017). Here, the focus is strictly on youth employment in food systems, while other types of interactions of youth with the food systems, for instance via consumption behavior, is out of scope of the analysis. Within the food system, employment opportunities include farming as well as non-farm occupations in the downstream segments of the food value chains including food processing, marketing and retail, as well as food-related services.

While the importance of agriculture and food systems in the majority of African economies, especially in rural areas, makes the distinction of employment opportunities into within and outside of the food system useful for analytical purposes, it has recently been asserted that this is a false dichotomy given the complexity of sectoral and spatial mobility patterns of the youth (HLPE, 2021). First, food systems and related employment extend beyond rural areas into towns and cities. Second, even within rural areas, the share of employment outside of the food systems is non-negligible; especially when secondary and tertiary activities are accounted for. Third, in line with the life-cycle effects, migration at young age to urban areas and out of agriculture does not necessarily need to be permanent, and a growing evidence on return migration suggests that it can also be conceptualized in terms of keeping options open before a full-time entrance into farming (Huijsmans et al., 2021). Fourth, even if migration is permanent, the linkages between mobility and food systems can remain strong, for instance via remittances used for productive investment in agriculture, or via seasonal labor migration to fill up labor shortages (HLPE, 2021).

Finally, although the focus of this literature review is on youth, youth outcomes cannot be analyzed in separation from childhood outcomes. The conditions prevailing during childhood, including health and nutrition, education, and economic shocks affect human capital and labor market outcomes as adults (Currie & Almond, 2011; McLeod & Kaiser, 2004; Currie & Thomas, 1999). Child labor, whether paid or unpaid, outside or within the household, as well as early marriage, affect human capital and occupational trajectories of individuals at later stages of life (Mussa et al., 2017; Mussa et al., 2018; Beegle, Dehejia & Gatti, 2009). These childhood-related aspects are therefore a necessary component of the framework in which to analyze youth employment. This is particularly relevant for designing policies which should encompass both short-term policies targeting current youth, but also long-term policies targeting those who will become future youths.

The conceptual framework presented above guides the analysis in this paper. The analysis is based on a literature review. The primary focus is on empirical studies published in peer-reviewed journals; however, since the academic literature on youth employment, especially in food systems, is nascent, grey literature is also included, provided that the publications contain information about the data and methods used. Search engines such as AgEcon and Google Scholar were used to identify relevant papers, which was then followed by the snowball method to find other publications. Guided by the conceptual framework, the resulting list of articles was reviewed based on their scope, data and methods, and sectors and countries of interest. With respect to the latter, while the main focus of the review is on the African countries, evidence from other regions of the world, especially from Asia which underwent similar demographic transition four-five decades ago (Resnick & Thurlow, 2015), is also used whenever lessons relevant for Africa can be drawn. Finally, to supplement the literature review, descriptive statistics based on the available secondary data are also included.

3 Macro-determinants of youth labor markets in Africa

3.1 Cohort size

Sub-Saharan Africa is expected to become the most populous region in the world by 2050, according to the recent projections; and to account for more than half of the growth of the world's population between 2019 and 2050, with an increase of more than one billion people over that period (UNDESA, 2019). Several countries, including Nigeria, DRC, Ethiopia and Tanzania, are among the countries with highest projected increase in population in absolute terms; while in relative terms, 18 countries in Sub-Saharan Africa are expected to double in size by 2050, with Niger even tripling in size (UNDESA, 2019). Sub-Saharan Africa is the only region where population is expected to continue growing throughout 2100, the latest point in population projections so far (UNDESA, 2019). These processes have straightforward implications for the region's age structure. With 62% of population below age 25 in 2019, Sub-Saharan Africa is the youngest region in the world. However, the share of population below age 25 has already started declining, albeit very slowly (Table 1).

The youth bulge, a common phenomenon in countries which have achieved a reduction in infant mortality but where fertility rate remains high, has therefore started receding in Sub-Saharan Africa (Thurlow, 2015). Instead, the adult working-age population (25 to 64) is the fastest growing population group in the region (UNDESA, 2019). The latter fact in particular creates potential for demographic dividend, that is a situation wherein the number of people in the workforce is higher than the number of dependents, and as such, presents opportunities for economic growth and structural transformation. Some authors consider, however, that the decline in fertility rates across Africa is too slow and might hamper the expected benefits from the demographic dividend (see Page, 2019). Nevertheless, while in absolute terms, the number of youths in Africa is very high and requires immediate policy action to create enabling environment for productive employment of current youths, the trends in relative terms suggest a slowly improving demographic structure on the continent.

	Youth sha	Youth share (%)		Historical peak share	
	2019	2050	%	Year	
World	21.0	17.4	29.7	1975	
Developing countries	22.8	18.3	33.0	1975	
Sub-Saharan Africa	34.4	28.2	36.0	2002	
Eastern Africa	35.7	27.1	37.2	2004	
Western Africa	34.6	29.6	35.8	2000	
Middle Africa	35.5	30.3	36.1	2005	
Southern Africa	24.3	19.3	33.0	1977	
excluding South Africa	29.9	22.2	37.5	2000	
Northern Africa and Western Asia	23.9	19.0	33.8	1978	
Central and Southern Asia	25.2	17.3	32.8	1977	
Eastern and Southeastern Asia	16.3	12.8	31.9	1973	
Latin America and the Caribbean	22.0	14.8	33.5	1976	

Table 1: Youth (15 - 24 years) share of potential labor force (15+ years)

2050 projections based on the medium fertility variant. Potential labor force measured as the total population aged 15 years and older.

Source: Author's own elaboration using World Population Prospects 2019 data from UNDESA. Table adapted from Thurlow (2015) and updated using recent population data.

3.2 Macroeconomic context

While African economy has been challenged by the impacts of the Covid-19 pandemic, with the continent's GDP contracting by 2.1% in 2020, for the first time in half a century (AfDB, 2021), it is projected to have quickly rebounded in 2021. Even though the rather optimistic future projections might be a sign of resilience of the African economy, it remains nevertheless vulnerable to global and regional shocks, and exposed to a number of important external and internal risks, including climate change and extreme weather events, political instability and social tensions, financial market volatility and debt overhang. Moreover, the Covid-19 pandemic has had important socio-economic impacts, reversing the trends of poverty reduction, with an estimated 30.4 million Africans pushed into poverty in 2020 and up to 38.7 million in 2021 (AfDB, 2021), and labor markets have been one of the main channels of these impacts. While in the formal sector, the reductions in employment were relatively small (Apedo-Amah et al., 2020; Baumüller et al., 2021), the impacts were disproportionately large in the informal sector, the main employer on the continent (Nguimkeu & Okou, 2020).

However, even before the shock of the pandemic, the solid economic growth in Africa did not translate into improvements on the local labor markets. The employment elasticity of growth remains very low; for instance, over the period 2000-2014, a 1% increase in GDP was associated with 0.41% increase in employment, and as a result, employment grew by 1.8% annually, much below the rate of growth of the labor force, at 3% (AfDB, 2019). Economic growth, while necessary, is not sufficient to boost employment in Africa, especially considering the structure of the economy. This "jobless growth" can partly be explained by the small contribution of the manufacturing sector to GDP; and by the fact that in many countries, growth is driven by natural resource extraction which is not labor-intensive. The estimates suggest that historically, industry-driven growth acceleration episodes exhibited three times higher employment elasticities compared to service-driven growth acceleration episodes in Africa (AfDB, 2019). In addition, the segmented labor markets and the prevalence of informality further hinder the positive spillovers from economic growth to employment.

3.3 Labor markets

Labor is the critical asset of the poor. Creating opportunities to leverage this asset and generate a steady income stream is the most sustainable pathway out of poverty. Labor force participation rate in Africa, at 63.1% and employment-to-population ratio, at 58.8 in 2019 (ILO, 2020), are higher than the global averages. This can be explained by the presence of large working-age population (see above) on the one hand, and lack of formal social protection and insurance mechanisms on the other. Most individuals in low income countries simply cannot afford to be idle (Filmer & Fox, 2014). In the context where formal unemployment insurance is scarce or absent, and where state regulations are weak, the relevance of quantity adjustment, as measured by the official unemployment rate, is very limited.

Instead, the key element of labor market supply and demand adjustment is in terms of employment quality (Herrera & Merceron, 2013). Only youth from the top end of income distribution, who often also happen to be highly educated, can afford to queue for good jobs (Filmer & Fox, 2014; Stampini & Verdier-Chouchane, 2011), whereas those who are too poor to withstand long unemployment spells have to either work at family farms or household enterprises or create their own jobs. In Sub-Saharan Africa, just 16% of work is in wage employment, divided roughly equally between public and private sector, with the former declining after the 1980s and 1990s reforms, with the exception of resource-rich countries (Filmer & Fox, 2014; Sumberg et al., 2012). The remaining 62% of population works in agriculture and 22% in household enterprises (Filmer & Fox, 2014), both identified as mainly informal sectors.

These sectors usually fall into the category of vulnerable employment, i.e. own-account and contributing family workers. In Sub-Saharan Africa, the problem of vulnerable employment affects around three-quarters of the employed population; and an estimated 70% of salaried workers in Sub-Saharan Africa do not have social protection (ILO, 2016). Informality, lack of contracts and social

security as well as poor working conditions are major issues even in cases of much of wage employment. Working poverty prevails. In 2018, 33% of workers in Africa lived in extreme poverty, and 22% in moderate poverty, despite having a job (ILO, 2019). Finally, the persistent market segmentation might inhibit optimal labor allocation, from the perspective of human capital, especially in rural areas. This evidence clearly indicates that in Africa, lack of adequate employment opportunities – which involve productive work and offer decent earnings, job security and social protection, among others – continues to be one of the principal challenges hindering poverty reduction efforts.

3.4 Entrepreneurship

Much of the earlier debate on labor markets in Africa focused on "efforts to expand formal wage employment opportunities and to promote structural transformation, out of lower productivity, subsistence agricultural activities into higher value-added employment in the services and industrial sectors" (ILO, 2014). This approach was later criticized as being too narrow to accommodate the rapidly growing working-age population in Sub-Saharan Africa. Projections suggest that over the next decade, more jobs will be created in household enterprises, i.e. around 45% of new jobs, particularly in self-employment (Fox & Thomas, 2016), rather than in wage employment. It is now recognized that self-employment, and, in particular, entrepreneurship – even though the definition of the latter and the distinction between the two remain unclear – can be the solution for the youth employment problem (AU, 2020).

Conceptually, the idea of entrepreneurship as a driving force of structural transformation and employment creation seems very attractive. For instance, Gries and Naudé (2010) propose an endogenous growth model with micro-economic foundations where entrepreneurs provide several roles in the economy: (i) create new firms outside of the household, (ii) absorb surplus labor from the traditional sector, (iii) provide innovative intermediate inputs to final goods producing firms, (iv) foster greater specialization in manufacturing, and ultimately (v) raise productivity and employment in both the modern and traditional sectors. Hausmann and Rodrik (2003) show that entrepreneurs provide technological externalities in bringing new goods to market, which also has information externalities and signaling function. However, the empirical evidence on the role of entrepreneurship in economic development is inconclusive and concerned with the micro-macro paradox (Hessels & Naudé, 2017), whereby firm-level studies show positive impacts of entrepreneurship in terms of job creation, investment, and technological progress, but where macro-level studies fail to show any positive relationship between entrepreneurship and economic growth (Gollin, 2008; Sautet, 2013).

Despite lack of clear evidence on entrepreneurship and, also, related policies targeting entrepreneurs (see, for example, Blattman, Dercon & Franklin, 2019; Fafchamps et al., 2014; Cho & Honorati, 2014), self-employment and entrepreneurship remain important in the debate on youth employment in Africa where youth cannot afford to queue for wage jobs, but instead have to create jobs for themselves (Filmer & Fox, 2014). In the short run, the biggest challenge is therefore to raise productivity of those self-employed, mainly in the informal sector. However, identifying self-employment with bad jobs might be misleading as there exists important heterogeneity of household enterprises (Teal, 2016; Fox & Sohnesen, 2012), in line with the informal sector's internal dualism and the two-tier distribution of jobs (Fields, 2009). Falco et al. (2011) show that size of the enterprise matters for the economic outcomes, including earnings. Although very often, informal enterprises are operated at such small scale that they turn out inefficient and become a poverty trap for the households (Banerjee & Duflo, 2007), larger enterprises, especially the ones with employees, offer earnings comparable or even higher than (some forms of) wage employment, irrespective of whether skills are controlled for or not (Teal, 2016; Kerr & Teal, 2015).

3.5 Mechanization and automation

Mechanization and automation in agriculture and industry also have the potential to shape labor markets in Africa. The empirical evidence for Africa is scarce, and has focused so far on mechanization in agriculture. Agricultural mechanization is seen as a way to increase labor productivity (Diao, McMillan & Mwangwe, 2018), in line with the theory of induced innovation (Hayami & Ruttan, 1985). Despite efforts to promote mechanization in agriculture first in the 1960s and 1970s (Pingali, 2007), and later after the 2008 food price crisis (FAO & UNIDO, 2008; Kienzle, Ashburner, & Sims, 2013; Mrema, Baker, & Kahan, 2008), food systems in Africa remain the least mechanized in the world (Pingali, 2007; Sheahan & Barrett, 2017); and governance challenges have been put forward as one of the explanations for these failed efforts (Daum & Birner, 2017). However, the recently emerging literature on the impacts of agricultural mechanization suggests that, along with increasing farmers' own income, mechanization can also increase the demand for hired labor when it leads to the expansion of cultivated land and when family labor is replaced with hired labor (Adu-Baffour, Daum & Birner, 2018). Similar findings emerge from other regions (Rajkhowa & Kubik, 2021).

On the other hand, the evidence on automation in Africa is largely unavailable, admittedly due to the low levels of industrialization and limited development of manufacturing across the continent. However, the emerging evidence from high-income or other developing regions of the world has spurred debate regarding the future impacts of automation in Africa as well (Kenny, 2019), to the extent that experiences from America or Asia are transferable to Africa. The available projections suggest that 47% of jobs in the US will be automatable within the next two decades, and that the share is even higher in developing economies (Frey & Osborne, 2017). Low-skilled workers in particular are at risk: Arntz, Gregory and Zierhan (2016) estimate that in the OECD countries, more than 50% of workers with primary (or lower) education, and 40% of those with lower secondary education are at risk of losing jobs to automation. Considering the concentration of employment in low-skilled jobs, demographic pressure and the already insufficient number of good jobs, such forecasts can be bad news for Africa (Kenny, 2019). Automation in manufacturing might also reverse the trends of locating low-skilled manufacturing and services in developing countries (ibid.). Irrespective of the future prospects, the production system of the fragmented global value chains has already been changing over recent years, with a declining job intensity of exports (Rodrik, 2018).

At the same time, the accuracy of these forecasts is highly debatable, especially that history has seen many episodes of such increased concerns with the impact of labor-saving technology – but so far, "the past two centuries of automation and technological progress has not made labor obsolete" (Autor, 2015). Autor (2015) suggests that while automation indeed substitutes for labor, it also complements labor, raises output in ways that lead to higher demand for labor, and interacts with adjustments in labor supply. Accordingly, in the case of Africa, Kenny (2019) argues that automation will be vital for development, productivity enhancement and creation of new jobs. The ongoing ICT revolution in particular might present new opportunities for the African youth. In this context, the demand for workers with specialized skillsets is expected to increase, and therefore it is necessary that education systems, including vocational training, adjust their curricula accordingly (HLPE, 2021).

4 Youth employment in food systems

4.1 Food systems and their potential

In this broader context, the potential of the food system to create jobs in Africa has been increasingly put forward in the literature (Allen et al., 2016; Townsend et al., 2017; Kubik et al., 2021). The African food market is projected to triple and reach US\$ 1 trillion by 2030 (WB, 2013), with the prospect of making agriculture and agribusiness a catalyst for development and poverty reduction. Growing food demand as well as food system transformation, both a consequence of rising per capita incomes, urbanization and new export opportunities, are expected to result in a shift of production from cereals to high value fresh, processed and convenience foods (ibid.). These changes in production patterns and related technologies will entail a higher level of food management and transformation beyond the farm, requiring major investments in processing, logistics, market infrastructure, and retail networks. The World Bank (WB) predicts that the most dynamic sectors are likely to be rice, feed grains, poultry, dairy, vegetable oils, horticulture and processed foods to supply domestic markets (WB, 2013).

Allen et al. (2016) emphasize that local food demand will shape employment patterns in the food system to a greater extent as compared to exports which will contribute to employment growth only in particular circumstances, especially for better-off farmers and agribusinesses, as is the case of, for example, horticulture in Kenya. Typically, perishable and processed foods for local markets require much value added, especially in the form of processing and packaging, as well as other services along the value chain, and their production is therefore highly labor-intensive. There is also a growing evidence of the role of local processing enterprises in supplying urban markets in Sub-Saharan Africa, where local firms dominate a range of product categories, be it in terms of number of products or number of firms (Ijumba et al., 2015; Snyder et al., 2015). Furthermore, contrary to popular perceptions, locally processed foods are widely present not only in traditional outlets, but also in modern retail chains (Allen et al., 2016).

A number of important constraints exist that might inhibit food systems from reaching their full potential. African agriculture still performs below its capacity, mainly as a result of insufficient investment, institutional and governance deficiencies, and lack of supportive hard and soft infrastructures (ZEF & Akademiya2063, 2020). In consequence, the development of agro-industries is hampered by poor vertical coordination with domestic farming and a limited ability of local farmers to consistently procure reliable quality raw materials. Many of the most dynamic agro-industries therefore still rely on raw material imports, especially in West Africa (Hollinger & Staatz, 2015). Earlier literature suggested that this situation contributed to a highly dualistic character of the agro-processing sector with a limited number of medium and large enterprises and a large number of small and micro operators, mostly in the informal sector – but without small and medium enterprises (SMEs) – "the missing middle" (Hollinger & Staatz, 2015; Dinh et al., 2012). However, more recent literature has provided evidence of rapid proliferation of SMEs in food systems across Africa (Reardon et al., 2019).

4.2 Food systems in youth aspirations

There is a common perception that youth are not interested in working in agriculture and food systems; and this perception is also supported by available evidence (Bezu & Holden, 2014; Maiga et al., 2015; Kosec et al., 2017; McCune et al., 2017). Leavy and Hossein (2014) call it a "generational break" whereby agriculture does not appear to be what the rural youth aspire to, despite the role it plays in local livelihoods. This apparent decline in interest in agriculture is influenced by the perceptions that young people have about agricultural work – as characterized by drudgery, exposure to shocks, low social status and poverty (Leavy & Hossein, 2014; Sruthi, Jayalal & Gopal, 2016). Structural barriers, and in particular lack of access to land, markets, capital and knowledge, are also prominent in shaping youth's aspirations related to agriculture (Bezu & Holden, 2014; Kosec et al.,

2017; Żmija et al., 2020). Young women in particular are discouraged from agricultural activities (Elias et al., 2018), potentially due to the gender-specific entry barriers.

These findings are corroborated by the data from a sms-survey covering 10,000 individuals between 18 and 35 years old from rural areas in 21 countries in Sub-Saharan Africa (henceforth sms-survey), which shows that among the most important job attributes, youth consider good wages (38% of respondents) and good working conditions (37.5%) as the most crucial. This is, to a great extent, in line with evidence from 32 countries in Africa, Asia, Europe, Latin America and the Caribbean suggesting that the majority of youth across all levels of education aspire to public-sector employment and high-skilled occupations; and that the most important factors are job security, formality and earnings (OECD, 2017). Finally, the sms-survey data shows that good quality of life is indicated by 8% of respondents as the most important job attribute; and this is also in line with Leavy and Hossein (2014) and White (2012) who point out that lack of interest in agriculture is often related to the low quality of life and lack of infrastructure in rural areas. Based on the same sms-survey, Melchers and Büchler (2017) list main factors which would make rural areas more attractive among rural youth: improvements to employment, education, technology, infrastructure, electricity and water, and support for agriculture.



Figure 2: Rural youth employment aspirations in selected countries in Africa

Source: ZEF and Akademiya2063 (2020) based on the GeoPoll dataset.

Against the evidence of youth losing interest in agriculture and food systems as potential career options, the sms-survey data shows that agriculture still occupies a prominent place in youth's employment aspirations. Overall, close to 25% of young Africans want to work in the food and

agriculture sector, but the share is higher in some countries, close to 40% in Kenya, Liberia, Malawi and Tanzania (Figure 2). The highest share of respondents, more than a third, want to work in services, communication and transport; and a quarter in the public sector. These figures, combined with relatively low importance of manufacturing, are in line with the discussion on structural transformation without industrialization (Gollin, 2018). Finally, LaRue et al. (2021) suggest that youth employment aspirations should be considered in the framework of complex livelihoods involving pluri-activity rather than a dichotomic choice between agricultural vs. non-agricultural jobs.

Moreover, other studies suggest that rather than having no aspirations to work in agriculture at all, these aspirations are conditional on selected factors, for instance availability of land or inputs, commercialization of agricultural activities, or combination of agriculture with other incomegenerating activities (White, 2020). Agriculture is also considered a viable option if it involves modern technology (Elias et al., 2018), or when it is perceived as a form of entrepreneurship – or agripreneurship for that matter (Bossenbroek, van der Ploeg & Zwarteveen, 2015; Mwaura, 2017). In this context, the role of education in changing perceptions about agriculture has been noted (Katz, 2004). This evidence is corroborated by the sms-survey data in Figure 3 which emphasizes the importance of technology, investment opportunities, and decent pay to make agriculture an attractive employment option. For 12% of respondents, the inadequate reputation of agriculture is a deciding factor. Only 3% of respondents consider that agriculture is not attractive at all. The results are broadly similar across female and male respondents.



Figure 3: Factors that make farming attractive according to rural youth in Africa

Source: Author's own elaboration based on the Geopoll dataset. Multiple answers allowed.

4.3 Status of youth employment in food systems

The food system is the dominant employer in Sub-Saharan Africa. Irrespective of aspirations, the share of youth actually involved in food systems in Africa is high, especially when looking at the time spent in agriculture compared to other forms of income-generating activities (Dolislager et al., 2021). Figure 4 presents various categories of employment in food systems: own farm self-employment, farm wage employment, agri-food system wage employment and self-employment, and other sectors wage and self-employment. The results are presented for several age categories (adults are also included for comparison) and several categories of rural areas, depending on their remoteness, and urban areas. The participation of youth in farming their own land is observed to be relatively high, declining with youth age categories, and then increasing again for adults, i.e. exhibiting a J-curve shape. In line with expectations, participation in farming is highest in the most remote rural areas and declines for less remote areas and locations in proximity to towns. Self-employment in the agri-food sector turns out to be the second most important employment category in rural areas for all age categories, whereas for older age categories and in areas located closer to towns self- and wage employment in other sectors is most prevalent. This high degree of involvement in agriculture puts downward pressure on the age structure of the sectoral workforce. Contrary to popular perceptions, the average age of African farmers is not rising – it is either stagnating or even falling in some countries (Yeboah & Jane, 2018). It is also much lower than previously claimed – not 60 years (FAO, 2014), but 34, once all the household members involved in family farming are considered rather than only the household heads (Arslan, 2019).



Figure 4: Time allocation by category of employment (share of full-time equivalents)



In some countries, employment in the off-farm segments of the food system is non-negligible and considerably contributes to the total employment in manufacturing sector. In Malawi and Tanzania, food and beverages account for more than 40% of total manufacturing employment (Townsend et al., 2017). Allen et al. (2016) show that in terms of job creation, off-farm food system is growing much more rapidly in percentage terms than farming, but since the growth is from a very small base, in absolute terms, the contribution of off-farm food system employment to new jobs is still lower than that of farming. At the moment, off-farm food system provides employment to less than 10% of youth who are still a much under-represented group in the business. On the other hand, off-farm food system tends to employ more women than men, especially in the sector of 'food away from home' where up to 90% of jobs are held by women (ibid.). However, when women participate in rural non-farm activities, they tend to concentrate in labor-intensive sectors rather than capital intensive ones, thereby trapped by low investment and low productivity rural non-farm enterprises (Haggblade et al., 2010).

Allen et al. (2016) indicate that despite these positive signs, growth of jobs in off-farm food system is still concentrated in wholesale and retail, rather than agro-processing. For example, in West Africa, marketing activities, i.e. transport, logistics, retail and wholesale, account for 68% of the off-farm food system employment, while food processing for only 22% (Allen, Heinrigs, & Heo, 2018). Allen et al. (2016) further elaborate that these employment patterns are the result of the important role that food imports still play in local consumption; while trading in imported grains and high value-added agricultural products can boost employment in wholesale and retail sectors, potential gains in the agro-processing are, to a great extent, still lost to international suppliers.

4.4 Labor productivity

In order for the food systems to attract youth, mainly in the rural areas, and provide them with viable employment, either as wage or self-employed, it is necessary to boost the productivity, both in farming and the off-farm segments of the food value chains (ZEF & Akademiya2063, 2020). Despite popular perceptions, on a per-hour basis, productivity in agriculture is not necessarily intrinsically lower than in other sectors (Hicks et al., 2017; Arthi et al., 2016); however, underemployment due to seasonality drives the average productivity figures down; on average, farmers in Sub-Saharan Africa work 26 hours a week, in comparison to 39 hours in non-farm activities (Allen, Heinrigs, & Heo, 2018). Townsend et al. (2017) and Allen, Heinrigs, and Heo (2018) suggest that the use of new technologies, irrigation, mechanization, mixed farming systems and diversification should lead to an increase in the annual productivity. In the context of the increasing weather risk for agriculture, climate-smart solutions can also help boost productivity and build resilience (Nyasimi et al., 2014). Additionally, these technologies, and in particular mechanization, are expected not only to enhance productivity, but also to change youth perceptions on certain negative aspects of agriculture such as drudgery. Also, even though it is still not clear how mechanization will affect farming labor demand, it might free up young farmers' time for off-farm income-generating activities (Yeboah, 2018; Biggs & Justice, 2015).

In line with Johnston and Mellor (1961), Johnston and Kilby (1975) and Mellor (1976), it is generally expected that productivity growth in agriculture will be the main catalyst of structural transformation in Sub-Saharan Africa, as it was the case in much of Asia. Over the last two decades, the highest non-farm labor productivity growth was observed in countries displaying the highest levels of agricultural productivity (Yeboah & Jayne, 2018). Therefore, employment strategies for youth "need to integrate a system approach that captures the links between agricultural productivity, off-farm employment and rural and urban areas" (Allen, Heinrigs & Heo, 2018). It is necessary to undertake initiatives boosting entrepreneurship and private sector participation, and improving the rural business climate. Townsend et al. (2017) suggest that this can be achieved by both macroeconomic measures, i.e. political and macroeconomic stability, promotion of local and international trade, or improvement in regulatory quality, as well as microeconomic initiatives, i.e. reduction of barriers to entry, or entrepreneurship programs.

Investments in rural infrastructure are of paramount importance, especially because in several sectors, like power generation, there has been no change over the last twenty years, with only around 35% of population having access to electricity, and rural access rates being less than one-third of urban ones (WB, 2017). Transport infrastructure is lagging behind, and road density has actually declined in Sub-Saharan Africa over this period (ibid.). If Sub-Saharan Africa closed the gap with what is considered by the World Bank as the regional norm, the GDP growth rate would increase by 1.7 p.p., i.e. 1.2 p.p. due to increased quantity, and 0.5 p.p. due to improved quality of infrastructure, with electric power showing the largest effects in terms of quantity, and roads in terms of quality (ibid.). Investing in infrastructure is important not only in that, in the long run, it facilitates production and enhances market access (Berg, Blankespoor, & Selod, 2016), but additionally, in the short run, it might be an opportunity for engaging youth in public works, and considered a temporary, yet potentially effective solution to the youth unemployment problem (von Braun & Kofol, 2017). Access to infrastructure in the rural areas might have also effects on rural youth aspirations, rural out-migration, and rural entrepreneurship.

ICT infrastructure in particular presents great opportunities. In telecommunications, Sub-Saharan Africa has seen a dramatic improvement (World Bank, 2017), and the expanding mobile networks coverage and user base as well as the 'Internet of Things' will enable advanced collection and sharing of data and knowledge (Baumüller, 2017); once again, not only potentially boosting productivity, but also changing perceptions of agriculture as a modern rather than backward sector. Several initiatives are already taking place in Sub-Saharan Africa. For example, AniTrack, a start-up based in Ghana, provides a radio-frequency identification device (RFID) that tracks cattle's health, and notifies the farmer if any inconsistencies occur (Yeboah, 2018). FarmerLine, another company based in Ghana, but also operating in other countries, offers production, market and weather information to farmers using short audio messages sent to basic cell phones (ibid.). Social media is also increasingly mobilized. Digital Farmers in Kenya is an example of how such platforms can serve as a virtual marketplace (ibid.). Even though promising, it remains unclear to which extent such digital solutions can be scaled up and reach a higher customer base, especially among the youth.

4.5 Mobility

Despite the potential that food systems present in terms of employment creation for the African youth, and despite due attention they have received over the last years, recent evidence suggests that the debate on youth employment should acknowledge the complexity of livelihoods of young people, and replace the standard narrative of dyadic choices between agriculture and non-agricultural jobs, or rural and urban livelihoods with a more comprehensive perspective of pluri-activity and pluri-locality (White, 2020; Rigg et al., 2020). The extent of multiple-job holdings, where individuals are engaged in several economic activities in parallel, is admittedly very high among young people; even though the conventional statistics are rarely able to capture these employment patterns. In particular, the standard labor force surveys exclude the possibility of combining school and work, be it part-time or unpaid (Bourdillon et al., 2020), and only report the main job. Living Standard Measurement Surveys (LSMS) offer some improvement by also recording the secondary activity; but the most accurate insights on youth employment can be gained from the time-use surveys which not only record economic and non-economic activities of individuals but also time spent on each activity (Daum et al., 2018; Dolislager et al., 2021). Unfortunately, due to the high monetary and time cost they require, such surveys are available only for a few countries.

Young people's livelihoods often include periods of mobility between places and sectors of economic activity (Rigg et al., 2020). Sumberg et al. (2012) point out that youths' livelihood choices are rarely permanent and thus, a life-course approach would be more appropriate to study their interactions with the food systems and rural spaces rather than static snapshots of employment patterns. For instance, in the sms-survey, more than 50% of respondents indicates that their rural-urban migration aspirations and prospects will depend on the conditions prevailing at the given point in time. For young

people, mobility out of agriculture does not necessarily preclude taking up farming at later stages in life (Huijsmans et al., 2021), especially considering the extent of circular and return migration. Second, out-migrants often retain a certain degree of engagement in farming back home through remittances that can be used for productive investment in agriculture (Rozelle, Taylor & deBrauw, 1999). "As such, the relationship between food systems, youth aspirations, employment and mobility is complex and does not follow a linear path" (HLPE, 2021).

5 Constraints

5.1 Access to land

Irrespective of their career and livelihood choices, young people face important barriers when entering labor markets – but also once they are already participating in the labor markets. The literature points out lack of or limited access to resources: knowledge and skills, finance, markets and land, as the principal constraints which young people are exposed to. It is often argued that these constraints are not exclusive to youth; and that adults are affected by the same constraints – which can also make the design of youth policies more difficult (Filmer & Fox, 2014). However, even if the obstacles are similar for young and adult workers in many contexts, the magnitude of impacts is usually much higher for youth due to the life-cycle effects. Because of their young age and dependency relationship with adults, especially parents, young people have limited or no control over resources – even those rightfully earned by them (Löwe & Phiona, 2017). These intergenerational power dynamics are very important when considering access to resources, as access to both tangible and intangible assets, including knowledge, is most often obtained through transfer between generations (HLPE, 2021).

Access to land is one such example. Land ownership is strongly concentrated among older adults (Filmer & Fox, 2014). In rural Sub-Saharan Africa, youth rely on inheritance and gifts, or informal rental markets, to access plots under usufruct land right systems (Kosec et al., 2017). However, the growing population pressure has led to severe land scarcity in many countries (Jayne, Mather, & Mghenyi, 2010). Even though, at the continental level, Africa is considered as land-abundant and the estimated 52% of the world's remaining arable land is in Africa (Deininger et al., 2011), most of this land is concentrated in eight countries, while the rest of countries face increasing land pressures (Chamberlain, Jayne & Headey, 2014). For example, in Ethiopia, the average farm size declined from 1.4 ha in 1977 to 1 ha in 2001, and in some regions, this figure is even lower, less than 0.5 ha (Headey, Dereje & Taffesse, 2014). The situation has deteriorated particularly for younger farmers: those below the age of 38 have farm sizes that are almost 0.2 ha smaller than farmers aged 50 years and about 0.3 hectare smaller than those aged 60 years (ibid.). In Ghana, young farmers' landholdings are typically 1 to 3 acres (Kidido, Bugri, & Kasanga, 2017). The problem lies not only in a decreasing total farm size – increasing land fragmentation is also an issue that has negative impacts on farm productivity (Ali & Deininger, 2015). Women are significantly disadvantaged with respect to their land rights (FAO, 2018).

Policies to improve access to land for young people are difficult to implement. Filmer and Fox (2014) argue that the two aspects of land administration that matter most to young people are improving land tenure security and relaxing controls on land rentals. Only 10% of occupied land in Africa is formally registered (WB, 2012), however, several countries, i.e. Rwanda, Ethiopia, Madagascar, Uganda and Ghana, have undertook the formal process of documenting individual land ownership (Filmer & Fox, 2014). It has been shown that the property rights documentation can reduce transaction costs related to property transfers; for example, in Ghana and Rwanda, these costs were reduced to less than 1% of the property value (WB, 2012). Land rights certification might positively affect land rental activity (Filmer & Fox, 2014) which, in turn, is expected to ease access to land for youth (Kosec et al., 2017). However, in many countries, for example in Ethiopia, the existing restrictions on land rental market continue to disadvantage prospective young farmers (Deininger, 2003). Therefore, removing restrictions on land markets would have the potential to create both efficiency and equity benefits (Deininger, Stavastano, & Xia, 2017). Finally, redistribution of agricultural land is another option that governments might consider in order to address the youth land issue; however, redistribution programs should be properly designed and attain a large enough scale to be effective. For example, Malawi's pilot land reform program seems to bring the expected results; on the other hand, South Africa's several attempts to redistribute land have failed so far (Filmer & Fox, 2014).

5.2 Access to finance

In 2014, only 20.5% of 15 – 24 years old adults held an account at a financial institution or through a mobile money provider; the respective share for older adults was 33%. Only 11% of African youth saved in a formal financial institution (Demirguc-Kunt et al., 2015). The worst situation is reported in the CFA system in West and Central Africa, where excessive regulation has restrained innovation in the financial sector and where the penetration of financial institutions is extremely low (Riquet & Mbenge, 2013). These figures do not imply that most of young adults in Sub-Saharan Africa do not save or contract loans – on the contrary, almost 50% reported saving some money but mostly in an informal way. Similarly, almost half of young adults reported contracting a loan but in only one out of fourteen cases it was a loan from a bank (Demirguc-Kunt et al., 2015). Access to finance remains one of the biggest constraints for youth to enter productive activities, especially in self-employment, whether on or off farm.

The challenges of financial expansion for youth in Sub-Saharan Africa are non-negligible. First of all, youth lack collateral: they rarely have formal land titles, valuable mobile assets, or a steady employment contract, which makes them ineligible for formal credit (Filmer & Fox, 2014). Additionally, since typically, credits contracted by youth are of a small amount, while the fixed costs for a bank are relatively large, the cost of lending to youth remains high. Even the most efficient financial institutions report that they have to earn on average 19% a year on their loan portfolios just to cover the operating costs (ibid.). Several innovations are proposed as a way to increase access to financial services. Alternative forms of collateral are one example; the 2010 amendment of the OHADA Uniform Act on Secured Transactions, effective in 17 Sub-Saharan countries, allows borrowers to use a variety of assets as collateral, i.e. warehouse receipts, movable property and receivables that remain in the hands of the debtor (AgriFin, 2012). Contracting arrangements are another option, and several already operating examples like Kenya's DrumNet Project seem promising (AGRA, 2015). Leasing, which requires less or even no collateral, could be a solution to acquire farm equipment, for example, DFCU Leasing in Uganda provided USD 4 million in farm equipment leases in 2002 (Kloeppinger-Todd, Nair & Mulder, 2004). Additionally, partial loan guarantees, like the one provided by the Alliance for a Green Revolution in Africa (AGRA) in Kenya, Mozambique and Tanzania, can result in lower interest rates (Filmer & Fox, 2014). None of these initiatives are targeted at youth specifically. Filmer and Fox (2014) argue that since youth are a particularly risky group, separating them from a larger risk-sharing pool would make them even less attractive to financial institutions. Proliferation of mobile finance and mobile banking across Africa has also been considered as a pathway to increase inclusiveness of financial services; however, Parlasca, Johnen and Qaim (2021) show that so far, mobile financial services have not yet had a transformative effect on agriculture, with fewer than 15% of farmers using them for agricultural purposes, and only 1% using mobile loans.

5.3 Skills and innovation

Sub-Saharan Africa has made a remarkable progress in educational participation and attainment. The share of children completing primary education increased from 54% in 1990 to 70% in 2016; important gains are also visible with respect to secondary education (Filmer & Fox, 2014). Between 2000 and 2017, the literacy rate of young people in Sub-Saharan Africa rose from 65.9% cent to 75.4% (ILO, 2020). However, the quality of education has not improved at the same pace. In a survey of experts from 36 African countries, 41% of respondents indicated lack of skills as a major obstacle for youth, while even a higher share reported a mismatch between skills held by job seekers and skills required by employers (AfDB, 2012). While the comparable data is largely missing, the three countries that participated in the Progress in International Reading Literacy Study (PIRLS) in 2016 (Egypt, Morocco and South Africa) were the three lowest ranked countries; and a similar outcome was reported for mathematics and science evaluated in 2015 by the Trends in International Mathematics and Science Study (TIMSS) (ILO, 2020).

In particular, there seems to be a disconnect between formal education systems and skill demand in the food systems despite the role they play in livelihoods and employment. So far, not only did formal education systems not have the capacity to prepare students for jobs in the agriculture and food sector, but, very often, they have perpetuated negative perceptions about these kinds of jobs among students (Katz, 2004; Ansell et al., 2020). Agriculture is often neglected in the tertiary education; only 2% of students specialize in agriculture – making it almost impossible for agro-industrial firms to find high-level managerial candidates in this sector, according to a large recruitment company operating in many African countries (AfDB, 2012). Even when they exist, the agricultural curricula in tertiary education, not only in Africa but also elsewhere in the world, tend to adopt very traditional approaches to teaching agriculture (Jordan et al., 2014) and are not adapted to the fast-changing realities of the food systems.

Formal education programs that relate to food systems need to provide a better tailored skillset to their graduates, also including soft skills like critical thinking, problem solving, collaboration or communication. They should offer trainings in new topics such as conservation agriculture, climate-smart agriculture, precision agriculture, organic farming or urban farming (UNEP, 2021, cited in HLPE, 2021); and equip students with skills required for professions such as agronomists, crop breeders, nutritionists, food entrepreneurs or policy advocates (Valley et al., 2018; HLPE, 2021).

Again, women are disadvantaged in the agriculture-related programs, especially at tertiary level, where their participation is significantly lower compared to men, even in countries where on average, women's participation in tertiary education is similar to that of men (Mukembo et al., 2017). In Sub-Saharan Africa, only 25% of students enrolled in post-secondary agricultural programs were women (Kruijssen, 2009), leading to very low numbers of female professionals in agriculture (Beintema & Di Marcantonio, 2019). Across agricultural research institutions of the Global South, women constitute only 20% of scientists (Beintema, 2006). In this context, systemic changes are required in educational systems, but also agricultural policy, in order to support participation of girls and women in employment in food systems (Glazebrook, Noll & Opoku, 2020). The importance of professional social networks has also been emphasized in this respect (Mukembo et al., 2014); the African Women in Agricultural Research and Development (AWARD) network being a good example.

In the short run, the shortfall in skills might be addressed with post-school training, i.e. apprenticeships and technical vocational education and training (TVET). However, currently, enrollment in TVET is very limited in Sub-Saharan Africa: only 4% of the 24 – 35 years old adults have ever attended this kind of training (Filmer & Fox, 2014). On the one hand, this might result from limited local capacities, on the other, since most TVET requires some secondary schooling, a big share of young people are excluded from taking up this opportunity. Besides, in case of agriculture, a recent review of TVET in selected Sub-Saharan countries suggests that the training which is offered too often does not match the needs of the farmers and other agribusiness actors (Kirui & Kozicka, 2018).

6 Policy considerations

The youth employment challenge clearly leaves a lot of room for policy intervention in order to address the barriers and constraints that young people in rural Africa face, and to foster their productive employment. The need for action has been increasingly recognized by the policymakers. In 2006, the African Union (AU) Member States adopted the African Youth Charter which sets a general framework for national youth policies. In 2008, the period of 2009 – 2018 was declared the African Youth Decade by the AU. These actions are considered to be an important milestone in the African youth policy process, as they require countries to commit to developing and implementing integrated cross-sectoral youth strategies (Gyimah-Brempong & Kimenyi, 2013). In 2014, out of 54 African countries, 23 had a national youth policy, and 14 were in the process of developing such a policy (Youth Policy, 2014).

Youth policies relative to employment include the following types of interventions: (1) training and skills development, (2) wage subsidies, (3) search and matching assistance, and (4) entrepreneurship promotion; or a combination of thereof (Kluve et al., 2017; McKenzie, 2017). The first three types of interventions can be classified as traditional interventions implemented under the umbrella of active labor market programs (ALMPs) with a focus on formal wage employment. Skills trainings operate on the labor supply side, and aim to increase the employability of workers through vocational training (McKenzie & Woodruff, 2014); wage subsidies are intended to increase the demand for labor, through subsidizing the cost of labor to firms; finally, search and matching assistance interventions aim to lower frictions that prevent demand from meeting supply in the labor market (McKenzie, 2017; Datta et al., 2018). Because formal jobs are often scarce in rural areas of developing countries, and the problem is further exacerbated by spatial mismatches in labor markets, with very different employment opportunities for the same skills depending on where individuals are located (McKenzie, 2017; Bryan, Chowdury & Mushfiq, 2014), it is often argued, as noted above, that youth employment policies should instead place emphasis on promoting entrepreneurship (Cho & Honorati, 2014).

Recent meta-analyses and systematic reviews of impact evaluations of labor market interventions targeting youth (Table 2) provide mixed evidence on their effectiveness in improving youth's labor market outcomes such as employment and earnings, and suggest that the effect is modest, if any. For instance, a meta-analysis by Kluve et al. (2017) indicates that only a third of the interventions included in the analysis had a positive and statistically significant impact on either of the two outcomes. However, it has to be noted that most of the available meta-analyses and reviews of ALMPs impact evaluations are, to a great extent, based on evidence from either high-income countries or, in case of low- and middle-income countries, on evidence from regions other than Africa (Table 2). At the same time, this evidence suggests that the effectiveness of ALMPs is usually higher in low- and middle-income countries compared to high-income countries. Also, even though the evidence from Africa is still limited compared to other regions, the literature has recently grown substantially. Table 3 lists the randomized control trial (RCT)-based impact evaluations of ALMPs in Africa published over the last ten years.

It is increasingly recognized that rather than the type of intervention, it is the design of the intervention as well as the local context that matter for the intervention success (Datta et al., 2018; Kluve et al., 2017; Puerto, 2007). This is also what makes comparable estimates of the effectiveness of various interventions difficult, since the interventions vary widely in terms of their objectives, characteristics, conditions and target groups. Also, Kluve et al. (2017) and Card, Kluve and Weber (2018) suggest that interventions take time to bring results. The latter study shows that ALMPs have relatively small average employment effects in the short run (1-3 p.p.) but larger average effects in the medium run (3-5 p.p.) and longer run (5-12 p.p.). The time profile of average impacts in the post-program period varies with the type of ALMP: job search assistance programs tend to have similar impacts in the short and long run; training and private sector employment programs have larger average effects in the medium and longer runs. Public sector employment subsidies tend to have small or even negative average impacts at all horizons (Card, Kluve & Weber, 2018). In contrast, a recent study by Blattman, Fiala and Martinez (2020) indicates that even though an intervention – Youth Opportunities

Programme in Uganda – raised participants' employment by 17% and earnings by 38% four years after the intervention (Blattman, Fiala, & Martinez, 2014); these gains have completely dissipated nine years after the intervention. Grantees' investment leveled off; individuals in the control group eventually increased their incomes through business and casual labor; and so, both groups converged in employment, earnings, and consumption levels. However, grants had lasting impacts on durable asset stocks and skilled work.

Hence, so far, no type of intervention appears to be more effective than the others (Kluve et al., 2017); even though some studies provide evidence in favor of job search assistance (Eichhorst & Rinne, 2015); or trainings (Card, Kluve & Weber, 2018). In addition, McKenzie (2017) emphasizes the need to also consider the cost-effectiveness of interventions. Here, trainings and skill development programs perform particularly bad; for instance, no training program included in the analysis of Blattman and Ralston (2015) passes the cost-benefit test. Therefore, there is clearly a need to tailor interventions to the needs and constraints faced by the actual target group; and also identify potential bottlenecks that might hamper the participants from reaping the full benefits of the interventions (Datta et al., 2018).

The evidence suggests that the effectiveness of employment interventions often varies by the target group. For instance, Card, Kluve and Weber (2018) show that across all types of the traditional ALMPs, i.e. skills trainings, job search assistance and wage subsidies, larger average effects are observed for females and long-term unemployed, and smaller average effects for older workers and youths. On the other hand, Cho and Honorati (2014) emphasize that among all target groups, entrepreneurship programs have a positive and large impact for youth. Additionally, in case of interventions focusing on youth employment specifically, a number of studies report gendered differences in outcomes. Among studies based on evidence from Africa, Cho et al. (2013), Fiala (2014) and Honorati (2015) point to better outcomes for males than for females; on the other hand, Abel, Burger and Piraino (2020) observe positive outcomes for women, but not for men. These differences might point to the existence of deeper structural barriers, especially for women, that hinder them from integrating labor markets and which traditional ALMPs are not able to overcome. In this context, it may be important to design programs focusing specifically on girls and women. The evidence available for selected African countries shows that indeed, such well-targeted and well-designed programs can have tremendous impacts on girls and women and their labor market outcomes (Blattman et al., 2013; Adoho et al., 2014; Bandiera et al., 2020). In particular, the findings in Bandiera et al. (2020) point to the fact that in case of female youth, it is important that, rather than being strictly limited to interventions relative to employment, programs should also incorporate life-skills likely to affect other aspects of female livelihoods, such as teen pregnancy, or early entry into marriage or cohabitation. Irrespective of the gender, several studies also suggest that a potential benefit of various youth employment interventions may be in how they can influence youth's self-confidence, aspirations and future perspectives (Cho et al., 2013; Abel, Burger & Piraino, 2020).

Moreover, youth employment policies in developing countries are mainly focused on increasing employability of youth – usually through training skills programs; but in the context where the supply of labor is completely elastic, as far as firms are concerned, – which, despite oversimplification, fits the reality of many countries in Sub-Saharan Africa – interventions such as vocational training, apprenticeships or internships will only increase the supply of eligible workers (Fox & Kaul, 2018; Datta et al., 2018). Teal (2016) shows that public policy in Africa has been much more successful at increasing the supply of education than the supply of jobs the educated youth want, creating a mismatch between the rapidly increasing supply of educated labor and the weak demand for such labor by firms. It is therefore necessary that policy-makers reconsider their youth employment policies and re-balance supply-side policies with demand-side policies. A review by Fox and Kaul (2018) suggests that among the demand-side interventions, which target aspects such as access to finance, financial literacy, business development services, tax policy, formalization, infrastructure, as well as minimum wages, subsidies and public works, the most effective are the ones that reduce constraints to firm expansion – which then encourages firms to hire more workers. Interventions targeted to larger firms are more likely to have substantial employment effects (Datta et al., 2018).

The evidence on job creation effects of the demand side interventions is less comprehensive and less systematic compared to the evidence on the supply side employment interventions (Datta et al., 2018). Since the interventions focus on firms and not workers, they are not necessarily targeted at jobs, even less so at jobs for youth; instead, they aim at increasing productivity and sales growth – however, the firms that generate sales and turnover are not always the same as the ones generating jobs (Mamburu, 2017). On the other hand, jobs can have positive externalities if, for instance, they reduce expenditures on social assistance programs, that increase the value of given jobs (Robalino & Walker, 2017). A recent review of interventions targeted to small and medium enterprises (SMEs) including a variety of services such as training, access to credit, support for innovation, export promotion, shows that programs can have positive effects on jobs, labor productivity, and revenue or profits but these effects are not large; the cost effectiveness of interventions is difficult to estimate (Cravo & Piza, 2016). On the other hand, sectoral approaches, especially those related to investment in value chains, appear to be highly correlated with job creation, as evidenced by the apparel industry in Bangladesh (Dinh et al., 2012) and coffee, garment and cut flower industries in Ethiopia (Ingraam & Oosterkamp, 2014).

While the evidence suggests that, if they are to be effective, youth employment policies should be comprehensive and include both demand- and supply-side interventions in parallel, in practice, and especially in poor countries where public budgets are subject to severe financial constraints, policymakers are forced to set priorities in implementation of interventions. As such, policy-makers are exposed to time-consistency problems whereby they tend to prioritize interventions with can potentially bring tangible results in the short run at the cost of neglecting the equally important longrun strategies of youth employment creation. This political economy aspect can, at least partially, explain why current youth employment policies focus so strongly on supply-side interventions while neglecting the demand-side interventions which take longer to realize. In addition, potential trade-offs between various policy measures also need to be accounted for. Measures that show complementarities in their impacts should be put forward – however, these aspects are still largely missing in the literature, where impact assessments of youth employment interventions are mostly done for specific interventions in separation from other potential interventions. Finally, from the policy perspective, it is important to ensure that a conducive legal framework and employment protection legislation (EPL) is put in place. Kuddo, Robalino and Weber (2015) stress the importance of regulatory constraints to hiring youth – in that they affect hiring and firing procedures as well as the cost of workers - even though these considerations are clearly more important for formal than informal employment; nevertheless, they can have important economy-wide implications as well.

In light of this evidence, it is again relevant to stress the potential of the food systems to create employment for youth in rural Africa. It has indeed been increasingly acknowledged by the policy-makers, for example in the framework of the Comprehensive Africa Agriculture Development Programme (CAADP) of the AU. However, despite formal commitments, only a few countries have integrated youth-specific programs in national agriculture and food security plans (AGRA, 2015). Ghana's National Youth Policy, one of the notable examples, aims to promote youth participation in modern agriculture as a viable career opportunity and a business option (Government of Ghana, 2010). Several actions are outlined, for example assisting youth in the formation of farmer organizations, facilitating access to credit by youth groups, advocating for the creation of land banks for youth or facilitating access to market (AGRA, 2015). Ghana's Ministry of Food and Agriculture, a major partner in implementing this strategy, devised Youth in Agriculture Program (YIAP), aiming at facilitating youth to undertake farming as a commercial venture by providing interest-free credit on innovative inputs and mechanization, or training in specific skills in food processing or sales (Mahama, 2012).

Several challenges still persist, especially with respect to policies implementation, resulting mainly from significant institutional gaps between the legislative and executive branches, limited political and financial power of the delegated youth ministries, or lack of capacity to undertake program monitoring and evaluation (AGRA, 2015). Furthermore, limited youth participation in the political process, especially with respect to the youth matters, is another obstacle for designing and implementing successful youth policies. In Sub-Saharan Africa, 80% of respondents in a global UN survey admitted

lack of opportunities for engagement in decision-making process within their communities (UN, 2012, cited in AGRA, 2015). However, the African Youth Charter clearly advocates for the right of youth to participate actively in the design, implementation and evaluation of development strategies and policies (AU, 2006). While national capacities often remain limited, a number of youth boards, associations, networks and other bodies operate at the international level across the Sub-Saharan Africa, for example Food, Agriculture and Natural Resources Policy Analysis Network (FANRPAN), Young Professionals for Agricultural Development (YPARD), Youth Alliance for Leadership and Development in Africa (YALDA), African Youth Initiative Network (AYINET) or Climate Smart Agriculture Youth Network (CSAYN) (AGRA, 2015). However useful and effective they are, they could not substitute for local youth groups which would enable a broader base of youth to voice their concerns and shape the decisions affecting their future.

Study	Methodology	Findings
META-ANALYSES		
Cho, Y., & Honorati, M. (2014).	A meta- analysis of impact estimates from	Entrepreneurship programs have a positive and large impact
Entrepreneurship programs in developing	37 impact evaluations of	for youth, among all target groups. The impacts are visible in
countries: A meta regression analysis. Labour	entrepreneurship programs. A wide range	terms of business knowledge and practice, but no
Economics, 28: 110-130.	of outcomes of interest include proxies of	immediate translation into business setup and expansion or
	labor market activities and labor market	increased income is observed. Providing a package of
	income and profits. Focus on studies	training and financing is most effective for labor activities.
	undertaken in developing countries;	Additionally, financing support appears more effective for
	studies based on the evidence from Africa	women and business training for existing entrepreneurs
	constitute 17% of the sample.	than other interventions to improve business performance.
		Involving the private sector for the delivery of programs and
		evaluating the program in the longer term appears to be
		more closely associated with improved effects of programs.
Eichhorst, W., & Rinne, U. (2015). An	A meta-analysis of impact estimates from	Robust impact assessment is missing for a majority of
Assessment of the Youth Employment	86 active labor market programs (ALMPs)	interventions in the YEI. For those with rigorous evaluation,
Inventory and Implications for Germany's	interventions targeting youth. A	the results suggest that youth employment interventions
Development Policy. I Z A Research Report No.	complementary qualitative assessment	are more effective in developing than developed countries.
67.	focuses on interventions in Youth	Job search assistance outperforms other types of
	Employment Inventory (YEI), most of them	interventions. Combined measures do not outperform
	implemented in developing countries.	programs that include only one type of intervention.
Kluve, J., Puerto, S., Robalino, D., Romero, J.M.,	A meta-analysis of impact estimates from	Youth employment interventions have a positive but small
Rother, F., Stöterau, J., Weidenkaff, F., & Witte,	113 experimental and quasi-experimental	effect on employment and earnings, no effect on business
M. (2017). Interventions to improve the labour	impact evaluations of training,	performance was demonstrated. The effects appear over
market outcomes of youth: A systematic review	entrepreneurship promotion,	the long run. Impacts are higher in low- and middle-income
of training, entrepreneurship promotion,	employment services and subsidized	countries compared to high-income countries. In low- and
employment services and subsidized	employment interventions targeting	middle-income countries, skills trainings and
employment interventions. Campbell	youth. Outcomes of interest include	entrepreneurship interventions produce the greatest
Systematic Reviews, 13: 1-288.	employment, earnings and business	impacts; interventions that include multiple components
	performance. Estimates based on the	lead to better outcomes. Rather than thy type of
	evidence from Africa constitute 10% of	intervention, the design matters more.
	the sample.	
Card, D., Kluve, J., & Weber, A. (2018). What	A meta-analysis of impact estimates from	ALMPs have relatively small average effects in the
works? A meta-analysis of recent active labor	207 impact evaluations of active labor	short run (1-3 p.p.) but larger average effects in

Table 2. Summary of recent meta-analyses and systematic reviews of labor market interventions targeting youth

market program evaluations. <i>Journal of the European Economic Association</i> , 16(3): 894-931.	market programs (ALMPs). Program effect measured in terms of probability of employment. Estimates based on the evidence from non-OECD countries constitute 15% of the sample.	the medium run (3-5 p.p.) and longer run (5-12 p.p.). Time profile of average impacts in the post-program period varies with the type of ALMP: job search assistance programs tend to have similar impacts in the short and long run; training and private sector employment programs have larger average effects in the medium and longer runs. Public sector employment subsidies tend to have small or even negative average impacts at all horizons. The average impacts of ALMPs vary across groups, with larger average effects for females and long-term unemployed, and smaller average effects for older workers and youths.
REVIEWS	1	T
McKenzie, D. (2017). How Effective Are Active Labor Market Policies in Developing Countries? A Critical Review of Recent Evidence, <i>World</i> <i>Bank Research Observer</i> , 32: 127-154.	Review of active labor market programs (ALMPs) impact evaluations in developing countries. The interventions are classified as (i) skill training, (ii) wage subsidies, and (ii) job search assistance.	Traditional ALMPs have little or no impact on employment and earnings; and their cost-effectiveness vary, with training programs found to be particularly costly. The most promising interventions appear to be ones that help workers access different labor markets, as well as overcome sectoral and spatial mismatches.
Datta et al. (2018). Integrated Youth Employment Programs. A Stocktake of Evidence on what works in Youth Employment programs. Jobs Working Paper, Issue No. 24, World Bank.	An extensive desk literature review and analysis of the major meta-analyses and literature reviews on both the labor demand side and labor supply side youth employment interventions.	Large body of evidence on the supply side youth employment interventions exists; but the evidence on the employment impacts of the demand side interventions is weak. Both the supply or demand side interventions are not automatically successful in creating jobs for youth (or any other groups), and design and implementation characteristics matter significantly. It is important to take an integrated approach because the long-term success of programs to improve employability of youth will depend on that there is demand for workers.

Source: Author's own elaboration.

Study	Country		Findings
SKILLS TRAININGS			
Honorati (2015)	Kenya	Kenya Youth Empowerment Project Intervention: classroom-based technical training coupled with three months of internships in private firms. Target: youth out of school and/or with no permanent job; with a minimum of eight years of education.	Positive impact on placing youths in paid jobs for male participants, with a 15% increase in probability of employment 15 months after the program. Positive effects on earnings were found only for older males, and females. No effects were found on the probability of starting up a new business, being self-employed, or working for a family business.
Adoho et al. (2014)	Liberia	Economic Empowerment of Adolescent Girls Intervention: technical and life-skills training with support to enter wage-employment or start a business. Target: Adolescent girls with basic literacy and numeracy skills, not enrolled in school.	The program increased employment by 47% and earnings by 80%. Positive effects on a variety of empowerment measures, including access to money, self-confidence, and anxiety about circumstances and the future were also documented.
Cho et al. (2013)	Malawi	Apprenticeship Training Programme and Entrepreneurial Support for Vulnerable Youth Intervention: apprenticeship programme. Target: vulnerable youth, i.e. orphans and school dropouts.	The program led to enhanced (self-reported) skills of the type that the training was intended to impart. Male trainees reacted by continuing to invest in their human capital development during the post-training period, but there were no significant effects on labor market outcomes in the short run. Better outcomes for boys compared to girls.
Bandiera et al. (2020)	Uganda	Empowerment and Livelihood for Adolescents Intervention: vocational and life-skills training; focus on skills related to self-employment and financial literacy; additionally, information on sex, reproduction, and marriage is provided. Target: adolescent girls.	Four years post intervention, adolescent girls in treated communities were more likely to be self-employed. Teen pregnancy, early entry into marriage/cohabitation, and the share of girls reporting sex against their will fell sharply.

Table 3: Summary of recent impact evaluations of youth employment interventions in Africa

WAGE SUBSIDIES			
Levinsohn et al. (2014)	South Africa	Youth Wage Subsidies for South Africa Intervention: a voucher for wage subsidy provided to individuals seeking jobs; firms that employ them are compensated for a portion of the wages; only applies to full-time formal employment. Target: youth in Johannesburg metropolitan area, eThekweni metropolitan area and Polokwane.	Participants who were allocated a wage subsidy voucher were more likely to be in wage employment both one year and two years after allocation. The impact of the voucher thus persisted even after it was no longer valid. The magnitude of these effects was relatively large – those in the voucher group were 7.4 p.p. (approximately 25%) more likely to be in wage employment one year after allocation and of similar magnitude two years later. This impact was not driven by changes in the sample composition.
JOB SEARCH ASSIST	ANCE		
Abebe et al. (2017a)	Ethiopia	Intervention: transport subsidy or job application workshop. Target: youth in Addis Ababa.	Despite finding large positive effects on job quality, no average treatment effects on the overall probability of having a job were detected. However, sizeable increase in earnings and employment rates among the most disadvantaged job-seekers were observed.
Abebe et al. (2017b)		Intervention: job fairs that match firms with a representative sample of young, educated job- seekers. Target: young unemployed people in Addis Ababa who completed high school.	The meetings at the fairs created very few jobs: one for approximately 10 firms that attended. This result can be explained by mismatched expectations: about wages, about firms' requirements, and the average quality of job-seekers.
Franklin (2017)	Ethiopia	Intervention: cash grant covering transport cost to the city center to receive information about new jobs. Target: young unemployed educated job- seekers.	Treated respondents increased job search intensity and were more likely to find good, permanent, jobs. Subsidies also induced a short- term reduction in temporary work. Cash constraints caused young people to give up looking for good jobs too early.
Bassi & Nansamba (2017)	Uganda	Intervention: (i) a matching component, whereby firms and workers are randomly matched for job interviews, and (ii) a signaling component, by introducing experimental variation in whether information on the soft skills of workers, such as work ethic and interpersonal skills, is disclosed to both sides of	Both workers and employers reacted to the information: managers of higher ability updated their beliefs on worker's skills, while workers with higher skills revised their outside options upwards, as proxied by their stated reservation wages and probability to be back in training one year post-intervention. The revelation of information on skills significantly increased overall job offers and hires for workers in the middle of the skill distribution: the probability

		the labor market, through the provision of certifications.	of employment at the matched firm increased from 2.3% to 5.5% for this group of workers.
		Target: young workers fresh out of vocational education and looking for jobs; and small and medium enterprises looking for workers.	
Abel, Burger &	South	Intervention: (1) applications with and without	(1) For the same applicant, attaching a letter increased the
Piraino (2020)	Africa	reference letters were submitted on behalf of	probability of receiving a response from 4.2% to 6.7% (a 60% increase)
		encouraged to obtain a reference letter	increase) The overall effect was driven by female applicants, whose
			response rates approximately doubled.
		Target: actual job seekers who visit the Labour	(2) Female participants who obtained letters were significantly more
		Centres.	likely to receive job interviews and to be employed after three
			months. On the other hand, no impacts were found for men.
ENTREPRENEURSH	IP PROMO	TION	1
Adoho et al.	Liberia	Economic Empowerment of Adolescent Girls	The program increased employment by 47% and earnings by 80%.
(2014)			Positive effects on a variety of empowerment measures, including
		Intervention: technical and life-skills training	access to money, self-confidence, and anxiety about circumstances
		start a husiness	and the future were also documented.
		Target: Adolescent girls with basic literacy and	
		numeracy skills, not enrolled in school.	
Premand et al.	Tunisia	Turning Theses into Enterprises	The entrepreneurship track was effective in increasing self-
(2012)			employment among applicants, but the effects are small in absolute
		Intervention: business training, including soft-	terms. In addition, the employment rate among participants
		skills; and personalized coaching, including	remained unchanged, pointing to a partial substitution from wage
		business plan development; seed capital for	employment to self-employment. The program fostered business
		best business plans.	skills, expanded networks, and affected a range of behavioral skills.
		Target: university students in applied	
		undergraduate curriculum.	
Fiala (2014)	Uganda	Start and Improve Your Business	Men with access to loans and training reported significantly higher
			profits. The loan-only intervention had some initial impact, but this
		Intervention: one of the following components:	did not last. There were no impacts from the grant intervention, and
		(i) a cash grant, (ii) a loan, (iii) business skills	no effects for women from any of the interventions. The results

		training and a loan, (iv) business skills training and a cash grant. Target: microenterprise owners in semi-urban	suggest that male-owned microenterprises can grow through finance when paired with training.
		locations.	
Blattman et al.	Uganda	Women's Income Generation Support	Dramatic increases in business startup and reduction in poverty were
(2013)			observed. A year after the intervention, monthly cash earnings
		Intervention: business skills training, cash grant,	doubled, savings tripled, and short-term expenditures and durable
		and support to start a non-farming business,	assets increased 30 to 50% relative to the control group. While the
		including business plan development. Approved	absolute changes seem small in magnitude, these were huge gains
		business plans eligible for start-up grant.	relative to where these women started. The treatment was most
			impactful on the people with the lowest initial levels of capital and
		Target: young women in war-affected region.	access to credit.
Blattman, Fiala &	Uganda	Youth Opportunities Programme	Four years on, an experimental evaluation found grants raised work
Martinez (2020)			by 17% and earnings by 38% (Blattman, Fiala, & Martinez, 2014).
		Intervention: one-time unsupervised cash	After nine years, these gains have dissipated. Grantees' investment
		grants, non-agricultural vocational training.	leveled off; controls eventually increased their incomes through
			business and casual labor; and so, both groups converged in
		Target: poor young people from small towns	employment, earnings, and consumption levels. However, grants had
		and villages in Uganda.	lasting impacts on durable asset stocks and skilled work.

Source: Author's own elaboration.

7 Conclusion

Demographic pressure and the youth bulge have spurred much debate in Africa. With the youngest population in the world and high numbers of new entrants into the labor market every year, Africa has to address its youth employment challenge. So far, despite sustained economic growth, African countries have not been successful in creating decent employment for their youth and absorbing the fast-growing labor force. The structure of their economies, heavily dependent on agriculture and, increasingly, on services, but with limited role of manufacturing, has resulted in very low employment elasticities of growth and jobless growth. Considering the already strenuous conditions in the local labor markets, characterized by high levels of informality, underemployment and working poverty, the projected influx of new entrants will therefore put additional pressure on the already insufficient number of decent jobs.

This paper provides a literature review on youth employment challenge in Africa. The literature on youth employment in Africa as well as in other regions of the world has been rapidly growing over the last years, building up available evidence that can inform policy. With the growing number of studies, the understanding of livelihoods, aspirations, opportunities and challenges of young people has substantially improved. Nevertheless, important gaps still persist, especially with respect to the impact assessment of youth employment policies and interventions, where the evidence form developing countries is still limited and often offering ambiguous findings. In addition, there is a growing recognition of complexity and non-linearity of youth employment transition trajectories which calls for adoption of comprehensive frameworks of analysis, in contrast to earlier linear or dyadic approaches. This also means that efforts to provide insights on the dynamics of youth transitions should complement the so far mostly static analyses.

The trajectories of youth transitions into productive employment are influenced by life-cycle effects, relative to the measure of accumulated life experience; and generational effects, relative to the measure of the conditions prevailing during an individual's formative years. A confluence of the two types of effects exposes young people to a number of constraints both when entering but also already once in the labor market. In case of rural Africa, with its many structural deficiencies and segmented labor markets which hinder optimal human capital allocation, such constraints are particularly important. These constraints have been conceptualized in the literature in terms of access to resources: skills and innovation, finance, and land. While these constraints are not unique for youth, as they also concern the adult population, they are often of greater magnitude or intensity in the case of youth. This is mainly due to the intergenerational power imbalances and limited agency of young people.

However, there are also reasons for cautious optimism. First, in the long run, the demographic structure in Africa will improve. The youth bulge has already started receding, while the adult workingage population is the fastest growing population group in the region. This creates potential for demographic dividend, and as such, presents opportunities for economic growth and structural transformation. To realize this potential, governments need to invest in youth and create enabling environment for their productive employment. While youth employment challenges have been increasingly recognized in policy at regional and country level, the actual measures implemented by governments tend to focus on supply-side interventions while leaving out demand-side interventions. Such approaches, especially in the context of Africa where labor supply is abundant, risk further enforcing imbalances in the labor market if the problem of insufficient labor demand is not addressed. Various labor market interventions, including training and skills development, wage subsidies, search and matching assistance and entrepreneurship promotions have been implemented in African countries. However, impact evaluations provide mixed evidence on their effectiveness in improving youth's labor market outcomes such as employment and earnings, and suggest that the effect is modest, if any. However, programs that target girls and women appear to have particularly strong impacts on their labor market outcomes.

The potential of food systems as catalysts for employment creation in Africa has been put forward in the literature. Growing food demand as well as food system transformation, both a consequence of rising per capita incomes, urbanization and new export opportunities, are expected to result in a shift of production from cereals to high value fresh, processed and convenience foods. This, in turn, is expected to translate into growing labor demand in the sector, not only in agriculture, but also in off-farm segments of food value chains which are usually considered to be highly labor-intensive. In terms of job creation, off-farm food system is growing much more rapidly in percentage terms than farming, but since the growth is from a very small base, in absolute terms, the contribution of off-farm food system employment to new jobs is still lower than that of farming. Self-employment in the agri-food sector is second, after farming, most dominant category of youth employment in rural areas.

Despite common perceptions that youth are not interested in working in agriculture and food systems, the data suggests that these jobs still play a significant role in the African youth's aspirations. Close to 25% of young Africans want to work in the food and agriculture sector, and the share is higher in some countries, close to 40% in Kenya, Liberia, Malawi and Tanzania. Accordingly, the average age of African farmers is not rising – it is also much lower than previously claimed, at 34 years of age and not 60. However, youth aspirations with respect to agriculture remain conditional on several factors that can make agriculture and food system jobs attractive, including technology, investment and market opportunities, and, of course, decent earnings. Increasingly, youth consider agriculture and other food system activities as a business venture for entrepreneurs. To fully realize its potential, however, systematic investments are necessary to increase labor productivity and incomes. Improvements in living conditions in rural areas are also vital if youth is to consider this form of employment. Finally, the literature has acknowledged the complexity of rural livelihoods, where pluri-activity, pluri-locality and various forms of mobility play an important role, creating strong linkages between sectors and between rural and urban spaces.

A limitation of this literature review is that is has not explicitly explored the diversity of contexts within the African continent. Lack of sufficient number of studies that would be adequately representative for regions or countries of Africa made it impossible to account for contextual factors susceptible of influencing local youth labor markets. Compared to most of Sub-Saharan Africa, the nature of youth employment challenge in Southern Africa and Norther Africa is very distinct. These differences should be further explored in order to advise locally relevant policies. Finally, it might be useful to have a closer look at lessons learned in Asia, where similar demographic process occurred earlier, and to see to which extent Africa can draw on Asian experiences.

References

- Abebe, G., Caria, S., Fafchamps, M., Falco, P., Franklin, S., & Quinn, S. (2017a). Anonymity of Distance? Job Search and Labour Market Exclusion in a Growing African City. SERC Discussion Paper 224.
- Abebe, G., Caria, S., Fafchamps, M., Falco, P., Franklin, S., Quinn, S., & Shilpi, F. (2017b). Matching Firms and Workers in a Field Experiment in Ethiopia. Policy Research Working Paper 8092, World Bank.
- Abel, M., Burger, R., & Piraino, P. (2020). The Value of Reference Letters: Experimental Evidence from South Africa. *American Economic Journal: Applied Economics*, 12(3): 40–71.
- Adoho, F., Chakravarty, S., Korkoyah, D.T., Lundberg, M., & Tasneem A. (2014). The Impact of an Adolescent Girls Employment Program. The EPAG Project in Liberia. Policy Research Working Paper 6832, World Bank.
- Adu-Baffour, F., Daum, T., & Birner, R. (2018). Can Big Companies' Initiatives to Promote Mechanization Benefit Small Farms in Africa? A Case Study from Zambia. ZEF – Discussion Papers on Development Policy No. 262, Center for Development Research, Bonn, June 2018.
- AfDB (2012). African Economic Outlook 2012. The African Development Bank, Abidjan.
- AfDB (2019). African Economic Outlook 2019. Macroeconomic Performance and Prospects. Jobs, Growth and Firm Dynamism. Integration for Africa's Economic Prosperity. African Development Bank, Abidjan.
- AfDB (2021). African Economic Outlook 2021. From Debt Resolution to Growth: the Road Ahead for Africa. African Development Bank, Abidjan.
- Afutu-Kotey, R. L., Gough, K. V., & Owusu, G. (2017). Young Entrepreneurs in the Mobile Telephony Sector in Ghana: From Necessities to Aspirations. Journal of African Business, 18(4), 476–491. https://doi.org/10.1080/15228916.2017.1339252
- AGRA (2015). Africa Agriculture Status Report 2015. Youth in Agriculture in Sub-Saharan Africa. Nairobi: Alliance for a Green Revolution in Africa.
- AgriFin (2012). Making Finance Work in Africa. Washington, DC: AgriFin.
- Ali, D.A., & Deininger. K. (2015). Is there a farm-size productivity relationship in African agriculture? Evidence from Rwanda. Land Economics, 91, 317-343.
- Allen. A., Howard, J., Kondo, M., Jamison, A., Jayne, T., Snyder, J., Tschirley, D., Yeboah, F.K. (2016). Agrifood youth employment and engagement study (AGYEES). Michigan State University.
- Allen, T., Heinrigs, P., & Heo, I. (2018). Agriculture, food and jobs in West Africa. West African Papers 14., OECD, Paris.
- Ansell, N., Froerer, P., Huijsmans, R., Dungey, C., Dost, A. & Piti. (2020). Educating "Surplus Population": Uses and Abuses of Aspiration in the Rural Peripheries of a Globalising World. Fennia – International Journal of Geography, 198(1–2): 17–38. https://doi.org/10.11143/fennia.90756
- Ansell, N. (2004). Secondary Schooling and Rural Youth Transitions in Lesotho and Zimbabwe. Youth & Society, 36, 183-202.
- Apedo-Amah, M.C., Avdiu, B., Cirera, X., Cruz, M., Davies, E., Grover, A., Iacovone, L., Kilinc, U., Medvedev, D., Maduko, F.O., Poupakis, S., Torres, J., Tran, T.T., 2020. Unmasking the Impact of COVID-19 on Businesses: Firm Level Evidence from Across the World. World Bank, Washington D.C.
- Arslan, A. (2019). How old is the average farmer in today's developing world? IFAD Blogs. URL https://www.ifad.org/en/web/latest/blog/asset/41207683 (accessed 8.22.20).

- Arntz, M., Gregory, T., & Zierahn, U. (2016). The Risk of Automation for Jobs in OECD Countries: A Comparative Analysis. OECD Social, Employment and Migration Working Papers No. 189. <u>https://doi.org/10.1787/5jl29h56dvq7-en</u>
- Arthi, V., Beegle, K, De Weerdt, J., & Palacios-Lopez, A. (2016). Not Your Average Job. Measuring Farm Labor in Tanzania. GLM LIC Working Paper 15. Bonn: IZA.
- Arulingam, I., Nigussie, L., Sellamuttu, S.S. & Debevec, L. (2019). Youth Participation in Small-Scale
 Fisheries, Aquaculture and Value Chains in Africa and the Asia-Pacific. FISH-2019-14. Penang,
 Malasia, CGIAR. pp. 1–66.
- Ashraf, A. (2016). Parental Aspirations and Schooling Investment: A Case of Rural Punjab, Pakistan. Pakistan Journal of Applied Economics, 26(2), 129–152.
- Autor, D.H. (2015). Why Are There Still So Many Jobs? The History and Future of Workplace Automation. Journal of Economic Perspectives, 29 (3), 3-30.
- Bandiera, O., Beuhren, N., Burgess, R., Goldstein, M., Gulesci, S., Rasul, I., & Sulaiman, M. (2020).
 Women's Empowerment in Action: Evidence from a Randomized Control Trial in Africa. *American Economic Journal: Applied Economics*, 12(1): 210–259.
- Banerjee, A.V., & Duflo, E. (2007). The Economic Lives of the Poor. Journal of Economic Perspectives, 21, 141-167.
- Bassi, V., & Nansamba, A. (2017). Information Frictions in the Labor Market Evidence from a Field Experiment in Uganda. IZA GLM LIC Working Paper No. 29.
- Baumüller, H., Kubik, Z., Dallimore, A., Getahun, T., & Velia, M. (2021). Impact of Covid-19 on Africa's food and beverage manufacturing companies: Evidence from selected African countries. Center for Development Research, University of Bonn, Bonn. Unpublished.
- Baumüller, H. (2017). Towards Smart Farming? Mobile Technology Trends and Their Potential for Developing Country Agriculture. In: Skouby, K.E., Williams, I., & Gyamfi, A. (eds.). Handbook for ICT in developing countries: 5G perspectives. Gistrup: River Publishers.
- Becker, G.S. (1965). A theory of the allocation of time. The Economic Journal, 75, 493-517.
- Beegle, K., Dehejia, R., & Gatti, R. (2009). Why Should We Care About Child Labor?: The Education, Labor Market, and Health Consequences of Child Labor. Journal of Human Resources, 44(4), 871– 889. <u>https://doi.org/10.3368/jhr.44.4.871</u>
- Beintema, N.M. (2006). Participation of Female Agricultural Scientists in Developing Countries. Brief prepared for the meeting "Women in Science: Meeting the Challenge," an adjunct to the CGIAR Annual General Meeting, Washington, DC.
- Beintema, N.M. & Di Marcantonio, F. (2019). Women's Participation in Agricultural Research and Higher Education – Key Trends in Sub-Saharan Africa. Washington, DC, and Nairobi, Kenya. International Food Policy Research Institute and CGIAR Gender and Diversity Program. (also available at https://doi.org/10.21955/gatesopenres.1116327.1).
- Berg, C.N., Blankespoor, B., & Selod, H. (2016). Roads and Rural Development in Sub-Saharan Africa. Policy Research Working Paper 7729. Washington, DC: The World Bank.
- Bernard, T., Dercon, S., & Taffesse, A. S. (2012). Beyond Fatalism: An Empirical Exploration of Self-Efficacy and Aspirations Failure in Ethiopia. ESSP Working Paper No. 46, International Food Policy Research Institute (IFPRI).
- Bezu, S., & Holden, S. (2014). Are Rural Youth in Ethiopia Abandoning Agriculture? World Development, 64, 259-272.
- Biggs, S., & Justice, S. (2015). Rural and Agricultural Mechanization: A Hisotry of the Spread of Small Engines in Selected Asian Countries. IFRPI Discussion Paper 1443. Washington, DC: International Food Policy Research Institute.
- Blattman, C., Fiala, N., & Martinez, S. (2020). The Long-Term Impacts of Grants on Poverty: Nine-Year Evidence from Uganda's Youth Opportunities Program. *AER: Insights*, 2(3): 287–304.

- Blattman, C., Dercon, S., & Franklin, S (2019). Impacts of Industrial and Entrepreneurial Jobs on Youth: 5-year Experimental Evidence on Factory Job Offers and Cash Grants in Ethiopia. NBER Working Paper 25788.
- Blattman, C., & Ralston, C. (2015). Generating Employment in Poor and Fragile States: Evidence from Labor Market and Entrepreneurship Programs. Mimeo.
- Blattman, C., Green, E., Annan, J., & Jamison, J. (2013). Building Women's Economic and Social Empowerment Through Enterprise. An Experimental Assessment of the Women's Income Generating Support (WINGS) Program in Uganda. LOGiCA Study Series No.1, World Bank.
- Bossenbroek, L., van der Ploeg, J.D. & Zwarteveen, M. (2015). Broken Dreams? Youth Experiences of Agrarian Change in Morocco's Saïss Region. Cahiers Agricultures, 24(6): 342–348. https://doi.org/10.1684/agr.2015.0776
- Bourdillon, M., Levison, D., Myers, W. & White, B. (2010). Rights and Wrongs of Children's Work. Rutgers University Press. (also available at <u>https://www.jstor.org/stable/j.ctt5hj7n8</u>).
- Bryan, G., Chowdury, S., & Mushfiq, A. (2014). Underinvestment in a Profitable Technology: The Case of Seasonal Migration in Bangladesh. *Econometrica*, 82(5):1671–1748.
- Card, D., Kluve, J., & Weber, A. (2018). What works? A meta-analysis of recent active labor market program evaluations. *Journal of the European Economic Association*, 16(3): 894-931.
- Chamberlin, J., Jayne, T.S., & Headey, D. (2014). Scarcity amid Abundance? Reassessing the potential for cropland expansion in Africa. Food Policy, 48, 51-65.
- Cho, Y. & Honorati, M. (2014). Entrepreneurship programs in developing countries: A meta regression analysis. Labour Economics, 28, 110-130.
- Cho, Y., Kalomba, D., Mobarak, A.M., & Orozco, V. (2013). Gender Differences in the Effects of Vocational Training: Constraints on Women and Drop-Out Behavior. IZA DP No. 7408.
- Cravo, T., & Piza, C. (2016). The Impact of Business Support Services for Small and Medium Enterprises on Firm Performance in Low -and Middle- Income Countries: A Meta-Analysis. Policy Research Working Paper 7664, World Bank.
- Currie, J., & Thomas, D. (1999). Early test scores, socioeconomic status and future outcomes. NBER Working Paper No. 6943.
- Currie, J., & Almond, D. (2011). Human capital development before age five. In Handbook of Labor Economics (Vol. 4, pp. 1315–1486).
- Datta et al. (2018). Integrated Youth Employment Programs. A Stocktake of Evidence on what works in Youth Employment programs. Jobs Working Paper, Issue No. 24, World Bank.
- Daum, T., & Birner, R. (2017). The neglected governance challenges of agricultural mechanisation in Africa–insights from Ghana. Food Security, 9(5), 959-979.
- Daum, T (2018). Of Bulls and Bulbs. Aspirations and perceptions of rural youth in Zambia. Unpublished working paper.
- Daum, T., Buchwald, H., & Birner, R. (2019). Times Have Changed: Using a Pictorial Smartphone App to Collect Time–Use Data in Rural Zambia. Field Methods, 31(1), 3-22. <u>https://doi.org/10.1177/1525822X18797303</u>
- Deininger, K. (2003). Land Policies for Growth and Poverty Reduction. Washington, DC: The World Bank.
- Deininger, K., Byerlee, D., Lindsay, J., Norton, A., Selod, H., & Stickler, M. (2011). Rising Global Interest in Farmland: Can It Yield Sustainable and Equitable Benefits? Washington, DC: The World Bank.
- Deininger, K., Savastano, S., & Xia, F. (2017). Smallholders' land access in Sub-Saharan Africa: A new landscape? Food Policy, 67, 78-92.

- Demirguc-Kunt, A., Klapper, L., Singer, D., & van Oudheusden, P. (2015). The Global Findex Database
 2014. Measuring Financial Inclusion around the World. Policy Research Working Paper 7255.
 Washington, DC: The World Bank.
- Diao, X., McMillan, M. S., & Mwangwe, S. (2018). Agricultural Labour Productivity and Industrialisation: Lessons for Africa. Journal of African Economies, 27(1). 28-65.
- Dinh, H.T., Palmade, V., Chandra, V., & Cossar, F. (2012). Light Manufacturing in Africa. Targeted Policies to Enhance Private Investment and Create Jobs. Washington, DC: The World Bank.
- Dolislager, M., Reardon, T., Arslan, A., Fox, L., Liverpool-Tasie, S., Sauer, C. & Tschirley, D.L. (2021).
 Youth and Adult Agrifood System Employment in Developing Regions: Rural (Peri-urban to Hinterland) vs. Urban, The Journal of Development Studies, 57(4), 571-593, DOI: 10.1080/00220388.2020.1808198
- Durham, D.L. (2017). Elusive Adulthoods: Introduction. Elusive Adulthoods: The Anthropology of New Maturities, pp. 1–210. Bloomington, USA, Indiana University Press.
- Eichhorst, W., & Rinne, U. (2015). An Assessment of the Youth Employment Inventory and Implications for Germany's Development Policy. I Z A Research Report No. 67.
- Elias, M., Mudege, N., Lopez, D.E., Najjar, D., Kandiwa, V., Luis, J., Yila, J. et al. (2018). Gendered Aspirations and Occupations Among Rural Youth, in Agriculture and Beyond: A Cross-regional Perspective. Journal of Gender, Agriculture and Food Security (Agri-Gender), 3(1): 82–107. https://doi.org/10.22004/ag.econ.293589
- Fafchamps, M., McKenzie, D., Quinn, S., & Woodruff, C. (2014). Microenterprise growth and the flypaper effect: Evidence from Ghana. Journal of Development Economics, 106, 211-226.
- Falco, P., Kerr, A., Rankin, N., Sandefur, J., & Teal, F. (2011). The returns to formality and informality in urban Africa. Labour Economics, 18, Supplement 1, S23-S31.
- FAO & UNIDO. (2008). Agricultural Mechanization in Africa. Time for Action. United Nations Industrial Development Organization and Food and Agriculture Organization, Vienna.
- FAO (2014). Contribution to the 2014 United Nations Economic and Social Council (ECOSOC) Integration Segment. Presented at the 014 ECOSOC Integration Segment, Food and Agriculture Organization of the United Nations, Rome.
- FAO (2018). The Gender Gap in Land Rights. Rome, FAO.
- Favara, M. (2017). Do Dreams Come True? Aspirations and Educational Attainments of Ethiopian Boys and Girls. Journal of African Economies, 26(5), 561–583.
- Fiala, N. (2014). Stimulating Microenterprise Growth: Results from a Loans, Grants and Training Experiment in Uganda. ILO.
- Fields, G.S. (2009). Segmented Labor Market Models in Developing Countries. In: Ross, D. & Kincaid, H. (Eds), The Oxford Handbook of Philosophy of Economics.
- Filmer, D., & Fox, L. (2014). Youth Employment in Sub-Saharan Africa. The World Bank. https://doi.org/10.1596/978-1-4648-0107-5
- Fox, L., & Sohnesen, T.P. (2012). Household Enterprises in Sub-Saharan Africa. Why They Matter for Growth, Jobs, and Livelihoods. Policy Research Working Paper 6184. The World Bank, Washington, DC.
- Fox, L., & Kaul, U. (2018). The Evidence is In. How Should Youth Employment Programs in Low-Income Countries be Designed? Policy Research Working Paper 8500, The World Bank, Washington D.C.
- Franklin, S. (2017). Location, Search Costs and Youth Unemployment: Experimental Evidence from Transport Subsidies. *The Economic Journal*, 128 (614): 2353-2379.
- Frey, C. B., & Osborne, M. A. (2017). The future of employment: How susceptible are jobs to computerisation? Technological Forecasting and Social Change, 114, 254–280. https://doi.org/10.1016/j.tech-fore.2016.08.019

- Glazebrook, T., Noll, S. & Opoku, E. 2020. Gender Matters: Climate Change, Gender Bias, and Women's Farming in the Global South and North. Agriculture, 10(7): 267. https://doi.org/10.3390/agriculture10070267
- Glover, D. & Sumberg, J. (2020). Youth and Food Systems Transformation. Frontiers in Sustainable Food Systems, 4. https://doi.org/10.3389/fsufs.2020.00101
- Gollin, D. (2008). Nobody's Business but my Own: Self-Employment and Small Enterprise in Economic Development. Journal of Monetary Economics, (55), 219–233.
- Gollin, D., 2018. Structural Transformation and Growth without Industrialisation (Background Paper Series No. 2). Pathways for Prosperity Commission, Oxford.
- Government of Ghana (2010). National Youth Policy of Ghana. Accra: Ministry of Youth and Sports.
- Gries, T. & Naudé, W. (2010). Entrepreneurship and Structural Economic Transformation. Small Business Economics Journal, (34), 13–29.
- Gyimah-Brempong, K., & Kimenyi, M.S. (2013). Youth policy and the future of African development. Africa Growth Initiative Working Paper 9. Washington, DC: Brookings.
- Haggblade, S., Hazell, P., & Reardon, T. (2010). The Rural Non-farm Economy: Prospects for Growth and Poverty Reduction. World Development, 38(10), 1429–1441. https://doi.org/10.1016/j.worlddev.2009.06.008
- Hausmann, R. & Rodrik, D. (2003). Economic Development as Self-Discovery. Journal of Development Economics, (72), 603–633.
- Hayami, Y., & Ruttan, V. (1985). Agricultural Development: An International Perspective. Johns Hopkins University Press, Baltimore, MD.
- Headey, D., Dereje, M., & Taffesse, A. S. (2014). Land constraints and agricultural intensification in Ethiopia: A village-level analysis of high-potential areas. Food Policy, 48, 129–141. https://doi.org/10.1016/j.foodpol.2014.01.008
- Herrera, J. & Merceron, S. (2013). Underemployment and Job Mismatch in Sub-Saharan Africa. In de Vreyer, P. & Roubaud, F. Urban Labor Markets in Sub-Saharan Africa. Washington, DC: The World Bank.
- Hessels, J. & Naudé, W. (2017). The Intersection of the Fields of Entrepreneurship and Development Economics: A Review towards a New View. IZA Discussion Paper No. 11103. IZA Institute for Labor Economics, Bonn, October 2017.
- Hicks, J.H., Kleemans, M., Li, N.Y., & Miguel, E. (2017). Reevaluating agricultural productivity gaps with longitudinal microdata. NBER Working Paper 23253.
- HLPE. 2017. Nutrition and Food Systems: A Report by the High Level Panel of Experts on Food Security and Nutrition of the Committee on World Food Security, Rome.
- HLPE. 2021. Promoting youth engagement and employment in agriculture and food systems. A report by the High Level Panel of Experts on Food Security and Nutrition of the Committee on World Food Security, Rome.
- Hollinger, F, & Staatz, J.M. (Eds.) (2015). Agricultural Growth in West Africa: Market and Policy Drivers. Rome: Food and Agriculture Organization.
- Honorati, M. (2015). The Impact of Private Sector Internship and Training on Urban Youth in Kenya. Policy Research Working Paper 7404, World Bank.
- Huijsmans, R. (Ed.) (2016). Generationing Development: An Introduction. pp. 1–31. https://doi.org/10.1057/978-1-137-55623-3_1
- Huijsmans, R., Ambarwati, A., Chazali, C. & Vijayabaskar, M. (2021). Farming, Gender and Aspirations
 Across Young People's Life-course: Attempting to Keep Things Open While Becoming a Farmer.
 European Journal of Development Research, 33(1).

- Ijumba, C., Snyder, J., Tschirley, D., & Reardon, T. (2015). Stages of Transformation in Food Processing and Marketing: Results of an Initial Inventory of Processed Foods in Dar es Salaam, Arusha, and Mwanza. Tanzania Policy Research Brief 3. Innovation Lab for Food Security.
- ILO (2016). World Employment and Social Outlook 2016. International Labour Office, Geneva.
- ILO (2019). The working poor or how a job is no guarantee of decent living conditions. International Labour Office, Geneva.
- ILO (2020). Report on employment in Africa (Re-Africa) tackling the youth employment challenge. International Labour Office, Geneva.
- Ingram, V., & Oosterkamp, E. (2014). Literature Review on the Labour Market Impacts of Value Chain Development Interventions. Wageningen, LEI Wageningen UR (University & Research centre), LEI Report.
- Jayne, T., Mather, D., & Mghenyi, E. (2010). Principal Challenges Confronting Smallholder Agriculture in Sub-Saharan Africa. World Development, 38, 1384-1398.
- Johnston, B.F., & Mellor, J.W. (1961). The Role of Agriculture in Economic Development. The American Economic Review, 51, 566-593.
- Johnston, B.F., & Kilby, P. (1975). Agriculture and Structural Transformation: Economic Strategies in Late Developing Countries. New York: Oxford University Press.
- Jordan, N., Grossman, J., Lawrence, P., Harmon, A., Dyer, W., Maxwell, B., Cadieux, K.V. et al. (2014). New Curricula for Undergraduate Food-Systems Education: A Sustainable Agriculture Education Perspective. NACTA Journal, 58(4): 302–310.
- Katz, C. 2004. Growing Up Global: Economic Restructuring and Children's Everyday Lives. Minneapolis, University of Minnesota Press.
- Kenny, C. (2019). Automation and AI Implications for African Development Prospects? CGD Note, Center for Global Development, Washington, October 2019.
- Kerr, A. and Teal, F. (2015). The determinants of earnings inequalities: Panel data evidence from KwaZulu-Natal, South Africa, Journal of African Economies, doi: 10.1093/jae/ejv006.
- Kidido, J.K., Bugri, J.T., & Kasanga, R.K. (2017). Dynamics of youth access to agricultural land under the customarytenure regime in the Techiman traditional area of Ghana. Land Use Policy, 60, 254-266.
- Kienzle, J., Ashburner, J. E., & Sims, B. G. (Eds.). (2013). Mechanization for Rural Development: A Review of Patterns and Progress from around the World (Vol. 20). Food and Agricultural Organization, Rome.
- Kirui, O. & Kozicka, M. (2018). Education and Training for Farmers and Other Actors in the Agri-Food Value Chain in Africa. ZEF Working Paper Series No. 164, Center for Development Research (ZEF), Bonn.
- Kloeppinger-Todd, R., Nair, J., & Mulder, A. (2004). Leasing: An Underutilized Tool in Rural Finance. ARD Discussion Paper 7, The World Bank, Washington D.C.
- Kluve, J., Puerto, S., Robalino, D., Romero, J.M., Rother, F., Stöterau, J., Weidenkaff, F., & Witte, M. (2017). Interventions to improve the labour market outcomes of youth: A systematic review of training, entrepreneurship promotion, employment services and subsidized employment interventions. *Campbell Systematic Reviews*, 13: 1-288.
- Kosec, K., Ghebru, H., Holtemeyer, B., Mueller, V., & Schmidt, E. (2017). The Effect of Land Access on Youth Employment and Migration Decisions: Evidence from Rural Ethiopia. American Journal of Agricultural Economics, 100, 931-954.
- Kruijssen, F. 2009. Youth Engagement in Agricultural Research: A Focus on Sub-Sahara Africa. Wageningen, Wageningen International. (also available at https://www.researchgate.net/publication/254894133_Youth_Engagement_in_Agricultural_Rese arch_A_Focus_on_Sub-Sahara_Africa).

- Kubik, Z., Odubote, I., Getahun, T., Oueslati-Zlaoui, M. (2021). Employment in agroprocessing sector:
 Evidence from Ethiopia, Ghana and Tunisia. Center for Development Research, University of Bonn,
 Bonn. Unpublished.
- Kuddo, A., Robalino, D., & and Weber, M. (2015). Balancing Regulations to Promote Jobs: from Employment Contracts to Unemployment Benefits. Washington, D.C.: World Bank.
- Langevang, T., Namatovu, R., & Dawa, S. (2012). Beyond necessity and opportunity entrepreneurship: motivations and aspirations of young entrepreneurs in Uganda. International Development Planning Review, 34(4), 439–459. <u>https://doi.org/10.3828/idpr.2012.26</u>
- LaRue, K., Daum, T., Mausch, K. & Harris, D. (2021). Who Wants to Farm? Answers Depend on How You Ask: A Case Study on Youth Aspirations in Kenya. The European Journal of Development Research: 1–25.
- Leavy, J. & Smith, S. (2010). Future Farmers: Youth Aspirations, Expectations and Life Choices. Brighton, Future Agricultures Consortium.
- Leavy, J. & Hossain, N. (2014). Who Wants to Farm? Youth Aspirations, Opportunities and Rising Food Prices. IDS Working Papers, 2014(439): 1–44. <u>https://doi.org/10.1111/j.2040-</u> 0209.2014.00439.x
- Levinsohn, J., Rankin, N., Roberts, G., & Schöer, V. (2014). Wage subsidies and youth employment in South Africa: Evidence from a randomized control trial. Stellenbosch Economic Working Papers: 02/14, University of Stellenbosch.
- Locke, C., & te Lintelo, D.J.H. (2012). Young Zambians 'Waiting' for Opportunities and 'Working Towards' Living Well: Lifecourse and Aspirations in Youth Transitions. Journal of International Development, 24, 777-794.
- Löwe, A., & Phiona, S. (2017). Creating opportunities for young people in Northern Uganda's agriculture sector. London: Overseas Development Institute.
- MacDonald, R. (2011). Youth Transitions, Unemployment and Underemployment. Journal of Sociology, 47, 427-444.
- Mahama, A.E.A. (2012). Youth Employment in Ghana: A Case Study in Ghana's Youth Africulture Programme. Mimeo.
- Maiga, E. Christiaensen, L., & Palacios-Lopez, A. (2015). Are the Youth Exiting Agriculture en Mass? Mimeo.
- Mamburu, M. (2017). Defining High-Growth Firms in South Africa. WIDER Working Paper 2017/107. Helsinki: UNU-WIDER
- Mausch, K., Harris, D., Heather, E., Jones, E., Yim, J., & Hauser, M. (2018). Households' aspirations for rural development through agriculture. Outlook on Agriculture, 47(2), 108–115. <u>https://doi.org/10.1177/0030727018766940</u>
- McCune, N., Rosset, P.M., Cruz Salazar, T., Morales, H. & Saldívar Moreno, A. (2017). The Long Road: Rural Youth, Farming and Agroecological Formacion in Central America. *Mind, Culture, and Activity*, 24(3): 183–198. <u>https://doi.org/10.1080/10749039.2017.1293690</u>
- McKenzie, D. (2017). How Effective Are Active Labor Market Policies in Developing Countries? A Critical Review of Recent Evidence, *World Bank Research Observer*, 32: 127-154.
- McKenzie, D., & Woodruff, C. (2014). What Are We Learning from Business Training Evaluations around the Developing World? *World Bank Research Observer*, 29 (1): 48–82.
- McLeod, J. D., & Kaiser, K. (2004). Childhood emotional and behavioral problems and educational attainment. *American Sociological Review*, 69(5), 636–658.
- Mekonnen, D. A., & Gerber, N. (2016). The effect of aspirations on agricultural innovations in rural Ethiopia (ZEF-Discussion Papers on Development Policy No. 214). Center for Development Research (ZEF), University of Bonn.

- Melchers, I. & Büchler, B. 2017. Africa's Rural Youth Speak Out. p. 3. (also available at https://www.rural21.com/fileadmin/downloads/2017/en-03/rural2017_03-S14-16.pdf).
- Mellor, J.W. (1976). The New Economics of Growth. Ithaca: Cornell University Press.
- Mgumia, J. H. (2017). Programme-Induced Entrepreneurship and Young People's Aspirations. IDS Bulletin-Institute of Development Studies, 48(3), 155–169. https://doi.org/10.19088/1968-2017.133
- Mrema, C. G., Baker, D., & Kahan, D. (2008). Agricultural Mechanization in Sub-Saharan Africa: Time for a New Look. Food and Agricultural Organization, Rome.
- Mukembo, S.C., Edwards, M.C., Ramsey, J.W. & Henneberry, S.R. (2014). Attracting Youth to Agriculture: The Career Interests of Young Farmers Club Members in Uganda. Journal of Agricultural Education, 55(5): 155–172.
- Mukembo, S.C., Uscanga, J.M., Edwards, M.C. & Brown, N.R. (2017). Increasing Female Enrollment for Agricultural Programs of Study in Sub-Saharan Africa: What Motivates Women to Pursue Careers in Agriculture? Journal of International Agricultural and Extension Education, 24(1): 17– 33. https://doi.org/10.5191/jiaee.2017.24104
- Mussa, E. C., Mirzabaev, A., & Admassie, A. (2017). Child labour: an Ethiopian perspective. Rural 21 The International Journal for Rural Development, (03/2017), 45–47.
- Mussa, E. C., Mirzabaev, A., Admassie, A., & Rukundo, E. N. (2018). Does childhood work impede long-term human capital accumulation? Empirical evidence from rural Ethiopia. International Journal of Educational Development. https://doi.org/10.1016/j.ijedudev.2018.09.001
- Mwaura, G.M. (2017). Just Farming? Neoliberal Subjectivities and Agricultural Livelihoods among Educated Youth in Kenya: Agricultural Livelihoods among Educated Youth in Kenya. Development and Change, 48(6): 1310–1335. https://doi.org/10.1111/dech.12342
- Nguimkeu, P. & Okou, C. (2020). A Tale of Africa Today: balancing the lives and livelihoods of informal workers during the COVID-19 pandemic. Policy Brief Issue 1, No. 3. Office of the Chief Economist, Africa Region, World Bank, October 2020.
- Nyasimi, M., Amwata, D., Hove, L., Kinyangi, J., & Wamukoya, G. (2014). Evidence of impact: Climatesmart agriculture in Africa'. CCAFS Working Paper no. 86. Copenhagen: CGIAR Research Program on Climate Change, Agriculture and Food Security (CCAFS).
- OECD (2017). Youth Aspirations and the Reality of Jobs in Developing Countries. Mind the Gap. The Organisation for Economic Co-operation and Development, Paris.
- Page, J. (2019). Harnessing Africa's Youth Dividend: A new approach for large-scale job creation. Brookings Institutions.
- Parlasca, M.C., Johnen, C., & Qaim, M. (2021). Use of mobile financial services among farmers in Africa: Insights from Kenya. Global Food Security, 32. https://doi.org/10.1016/j.gfs.2021.100590
- Pingali, P. (2007). Agricultural Mechanization: Adoption Patterns and Economic Impact. In R. Evenson & P. Pingali (Eds.), Handbook of Agricultural Economics, Volume 3 (pp. 2779–2805). North-Holland.
- Premand, P., Brodmann, S., Almeida, R., Grun, R., & Barouni, M. (2012). Entrepreneurship Training and Self-Employment among University Graduates. Evidence from a Randomized Trial In Tunisia. Policy Research Working Paper 6285, World Bank.
- Rajkhowa, P. & Kubik, Z. (2021). Revisiting the relationship between farm mechanization and labour requirement in India. Indian Economic Review. <u>https://doi.org/10.1007/s41775-021-00120-x</u>
- Reardon, T., Awokuse, T., Haggblade, S., Kapuya, T., Liverpool-Tasie, S., Meyer, F., Minten, B., Nyange, D., Rusike, J., Tschirley, D., & Vos, R. (2019). Quiet Revolution and Emerging Modern Revolution in Agri-food Processing in Sub-Saharan Africa. In: AGRA, Africa Agriculture Status Report: The Hidden Middle: A Quiet Revolution in the Private Sector Driving Agricultural Transformation (Issue 7). Alliance for a Green Revolution in Africa, Nairobi.

- Resnick, D. & Thurlow, J. (2015). African Youth at a crossroads. In: Resnick, D. & Thurlow, J. (Eds.), African Youth and the Persistence of Marginalization. Routledge: London.
- Rigg, J., Phongsiri, M., Promphakping, B., Salamanca, A. & Sripun, M. (2020). Who Will Tend the Farm? Interrogating the Ageing Asian Farmer. The Journal of Peasant Studies, 47(2): 306–325. <u>https://doi.org/10.1080/03066150.2019.1572605</u>
- Riquet, C., & Mbenge, D.M. (2013). Deepening Financial Inclusion in West Africa. Consultative Group to Assist the Poor.
- Robalino, D., & Walker, D.I. (2017). Economic Analysis of Jobs Investment Projects. Jobs Working Paper No. 7, World Bank.
- Rodrik, D. (2018). New Technologies, Global Value Chains, and Developing Economies. NBER Working Paper No. 25164. https://doi.org/10.3386/w25164
- Rozelle, S., Taylor, J.E., & deBrauw, A. (1999). Migration, Remittances, and Agricultural Productivity in China. The American Economic Review, 89: 2, Papers and Proceedings of the One Hundred Eleventh Annual Meeting of the American Economic Association (May, 1999), 287-291.
- Sautet, F. (2013). Local and Systemic Entrepreneurship: Solving the Puzzle of Entrepreneurship and Economic Development. Entrepreneurship Theory and Policy, (March), 387–402.
- Sheahan, M., & Barrett, C. B. (2017). Ten striking facts about agricultural input use in Sub-Saharan Africa. Food Policy, 67, 12–25. <u>https://doi.org/10.1016/j.foodpol.2016.09.010</u>
- Snyder, J., Ijumba, C., Tschirley, D., & Reardon, T. (2015). Local Response to the Rapid Rise in Demand for Processed and Perishable Foods: Results of an Inventory of Processed Food Products in Dar es Salaam. Tanzania Policy Research Brief 2. Innovation Lab for Food Security Policy.
- Sruthi P, Jayalal, L. & Gopal, N. (2016). Gender Roles in Fisheries along the Vembanad Estuarine System. Asian Fisheries Science, (29S): 193–203.
- Stampini, M., & Verdier-Chouchane, A. (2011). Labor Market Dynamics in Tunisia: The Issue of Youth Unemployment. IZA Discussion Paper 5611. IZA Institute for the Study of Labour, Bonn.
- Sumberg, J., Anyidoho, N.A., Leavy, J., te Lintelo, D.J.H. & Wellard, K. (2012). Introduction: The Young People and Agriculture "Problem" in Africa. IDS Bulletin, 43(6): 1–8. https://doi.org/10.1111/j.1759-5436.2012.00374.x
- Tanguy, B., Dercon, S., Orkin, K., & Taffesse, A. S. (2014). The Future in Mind: Aspirations and Forward-Looking Behaviour in Rural Ethiopia. SSRN Electronic Journal. https://doi.org/10.2139/ssrn.2514590
- Teal, F. (2016). Policies for Job Creation in Poor Countries. GLM LIC Working Paper 12, IZA Institute for Labour Economics, Bonn.
- Thurlow, J. (2015). Youth employment prospects in Africa. In: Resnick, D. & Thurlow, J. (Eds.), African Youth and the Persistence of Marginalization. Routledge: London.
- Todaro, M.P. (1969). A model of labor migration and urban unemployment in less developed countries. The American Economic Review, 59, 138-148.
- Townsend, R., Benfica, R., Prasann, A., & Lee, M. (2017). Future of Food. Shaping the Food System to Deliver Jobs. Washington, DC: The World Bank.
- UNDESA (2019). World Population Prospects 2019: Highlights. United Nations, Department of Economic and Social Affairs, Population Division (ST/ESA/SER.A/423), New York.
- Valley, W., Wittman, H., Jordan, N., Ahmed, S. & Galt, R. (2018). An Emerging Signature Pedagogy for Sustainable Food Systems Education. Renewable Agriculture and Food Systems, 33(5): 467–480. https://doi.org/10.1017/S1742170517000199
- von Braun, J., & Kofol, C. (2017). Expanding Youth Employment in the Arab Region and Africa. ZEF Working Paper 155. Bonn: Center for Development Research.
- WB (2012). Land Administration and Reform in Sub-Saharan Africa: From Piloting to Scaling Up. Washington, DC: The World Bank.

- WB (2013). Growing Africa: Unlocking the Potential of Agribusiness. Growing Africa : Unlocking the Potential of Agribusiness
- WB (2017). Africa's Pulse. An analysis of issues shaping Africa's economic future. Washington, DC: The World Bank.
- White, B. (2012). Agriculture and the Generation Problem: Rural Youth, Employment and the Future of Farming. IDS Bulletin, 43(6): 9–19. <u>https://doi.org/10.1111/j.1759-5436.2012.00375.x</u>
- White, B. (2020). Rural Household Pluriactivity and Plurilocality: A Source of Resilience to Climate Breakdown. IOP Conference Series: Earth and Environmental Science, 451: 012001. <u>https://doi.org/10.1088/1755-1315/451/1/012001</u>
- Yeboah, F.K., Jayne, T.S. (2018). Africa's Evolving Employment Trends. The Journal of Development Studies 54, 803–832.
- Yeboah, K.T. (2018). Youth for Growth: Transforming Economies through Agriculture. Chicago: The Chicago Council on Global Affairs.

Youth Policy (2014). The state of youth policy in 2014. Youth Policy Press, Berlin.

ZEF & Akademiya2063 (2020). From Potentials to Reality: Transforming Africa's Food Production. Investment and policy priorities for sufficient, nutritious and sustainable food supplies. Center for Development Research, Bonn, October 2020.

Żmija, K., Fortes, A., Nganwani, M., Sandra, T., Sylvester, Š., Ayambila, N., Żmija, D. et al. (2020). Small Farming and Generational Renewal in the Context of Food Security Challenges. Global Food Security, 26: 100412. <u>https://doi.org/10.1016/j.gfs.2020.100412</u>. ZEF Working Paper Series, ISSN 1864-6638

Center for Development Research, University of Bonn

Editors: Christian Borgemeister, Joachim von Braun, Manfred Denich, Till Stellmacher and Eva Youkhana

- **1.** Evers, Hans-Dieter and Solvay Gerke (2005). Closing the Digital Divide: Southeast Asia's Path Towards a Knowledge Society.
- **2.** Bhuiyan, Shajahan and Hans-Dieter Evers (2005). Social Capital and Sustainable Development: Theories and Concepts.
- **3.** Schetter, Conrad (2005). Ethnicity and the Political Reconstruction of Afghanistan.
- **4.** Kassahun, Samson (2005). Social Capital and Community Efficacy. In Poor Localities of Addis Ababa Ethiopia.
- **5.** Fuest, Veronika (2005). Policies, Practices and Outcomes of Demand-oriented Community Water Supply in Ghana: The National Community Water and Sanitation Programme 1994 2004.
- **6.** Menkhoff, Thomas and Hans-Dieter Evers (2005). Strategic Groups in a Knowledge Society: Knowledge Elites as Drivers of Biotechnology Development in Singapore.
- 7. Mollinga, Peter P. (2005). The Water Resources Policy Process in India: Centralisation, Polarisation and New Demands on Governance.
- 8. Evers, Hans-Dieter (2005). Wissen ist Macht: Experten als Strategische Gruppe.
- **8.a** Evers, Hans-Dieter and Solvay Gerke (2005). Knowledge is Power: Experts as Strategic Group.
- **9.** Fuest, Veronika (2005). Partnerschaft, Patronage oder Paternalismus? Eine empirische Analyse der Praxis universitärer Forschungskooperation mit Entwicklungsländern.
- **10.** Laube, Wolfram (2005). Promise and Perils of Water Reform: Perspectives from Northern Ghana.
- **11.** Mollinga, Peter P. (2004). Sleeping with the Enemy: Dichotomies and Polarisation in Indian Policy Debates on the Environmental and Social Effects of Irrigation.
- **12.** Wall, Caleb (2006). Knowledge for Development: Local and External Knowledge in Development Research.
- **13.** Laube, Wolfram and Eva Youkhana (2006). Cultural, Socio-Economic and Political Con-straints for Virtual Water Trade: Perspectives from the Volta Basin, West Africa.
- **14.** Hornidge, Anna-Katharina (2006). Singapore: The Knowledge-Hub in the Straits of Malacca.
- **15.** Evers, Hans-Dieter and Caleb Wall (2006). Knowledge Loss: Managing Local Knowledge in Rural Uzbekistan.
- **16.** Youkhana, Eva; Lautze, J. and B. Barry (2006). Changing Interfaces in Volta Basin Water Management: Customary, National and Transboundary.
- **17.** Evers, Hans-Dieter and Solvay Gerke (2006). The Strategic Importance of the Straits of Malacca for World Trade and Regional Development.
- **18.** Hornidge, Anna-Katharina (2006). Defining Knowledge in Germany and Singapore: Do the Country-Specific Definitions of Knowledge Converge?
- **19.** Mollinga, Peter M. (2007). Water Policy Water Politics: Social Engineering and Strategic Action in Water Sector Reform.
- **20.** Evers, Hans-Dieter and Anna-Katharina Hornidge (2007). Knowledge Hubs Along the Straits of Malacca.
- **21.** Sultana, Nayeem (2007). Trans-National Identities, Modes of Networking and Integration in a Multi-Cultural Society. A Study of Migrant Bangladeshis in Peninsular Malaysia.
- **22.** Yalcin, Resul and Peter M. Mollinga (2007). Institutional Transformation in Uzbekistan's Agricultural and Water Resources Administration: The Creation of a New Bureaucracy.
- **23.** Menkhoff, T.; Loh, P. H. M.; Chua, S. B.; Evers, H.-D. and Chay Yue Wah (2007). Riau Vegetables for Singapore Consumers: A Collaborative Knowledge-Transfer Project Across the Straits of Malacca.
- **24.** Evers, Hans-Dieter and Solvay Gerke (2007). Social and Cultural Dimensions of Market Expansion.
- **25.** Obeng, G. Y.; Evers, H.-D.; Akuffo, F. O., Braimah, I. and A. Brew-Hammond (2007). Solar PV Rural Electrification and Energy-Poverty Assessment in Ghana: A Principal Component Analysis.
- **26.** Eguavoen, Irit; E. Youkhana (2008). Small Towns Face Big Challenge. The Management of Piped Systems after the Water Sector Reform in Ghana.
- **27.** Evers, Hans-Dieter (2008). Knowledge Hubs and Knowledge Clusters: Designing a Knowledge Architecture for Development

- **28.** Ampomah, Ben Y.; Adjei, B. and E. Youkhana (2008). The Transboundary Water Resources Management Regime of the Volta Basin.
- **29.** Saravanan.V.S.; McDonald, Geoffrey T. and Peter P. Mollinga (2008). Critical Review of Integrated Water Resources Management: Moving Beyond Polarised Discourse.
- **30.** Laube, Wolfram; Awo, Martha and Benjamin Schraven (2008). Erratic Rains and Erratic Markets: Environmental change, economic globalisation and the expansion of shallow groundwater irrigation in West Africa.
- **31.** Mollinga, Peter P. (2008). For a Political Sociology of Water Resources Management.
- **32.** Hauck, Jennifer; Youkhana, Eva (2008). Histories of water and fisheries management in Northern Ghana.
- **33.** Mollinga, Peter P. (2008). The Rational Organisation of Dissent. Boundary concepts, boundary objects and boundary settings in the interdisciplinary study of natural resources management.
- **34.** Evers, Hans-Dieter; Gerke, Solvay (2009). Strategic Group Analysis.
- **35.** Evers, Hans-Dieter; Benedikter, Simon (2009). Strategic Group Formation in the Mekong Delta The Development of a Modern Hydraulic Society.
- **36.** Obeng, George Yaw; Evers, Hans-Dieter (2009). Solar PV Rural Electrification and Energy-Poverty: A Review and Conceptual Framework With Reference to Ghana.
- **37.** Scholtes, Fabian (2009). Analysing and explaining power in a capability perspective.
- **38.** Eguavoen, Irit (2009). The Acquisition of Water Storage Facilities in the Abay River Basin, Ethiopia.
- **39.** Hornidge, Anna-Katharina; Mehmood Ul Hassan; Mollinga, Peter P. (2009). 'Follow the Innovation' A joint experimentation and learning approach to transdisciplinary innovation research.
- **40.** Scholtes, Fabian (2009). How does moral knowledge matter in development practice, and how can it be researched?
- **41.** Laube, Wolfram (2009). Creative Bureaucracy: Balancing power in irrigation administration in northern Ghana.
- **42.** Laube, Wolfram (2009). Changing the Course of History? Implementing water reforms in Ghana and South Africa.
- **43.** Scholtes, Fabian (2009). Status quo and prospects of smallholders in the Brazilian sugarcane and ethanol sector: Lessons for development and poverty reduction.
- **44.** Evers, Hans-Dieter; Genschick, Sven; Schraven, Benjamin (2009). Constructing Epistemic Landscapes: Methods of GIS-Based Mapping.
- **45.** Saravanan V.S. (2009). Integration of Policies in Framing Water Management Problem: Analysing Policy Processes using a Bayesian Network.
- **46.** Saravanan V.S. (2009). Dancing to the Tune of Democracy: Agents Negotiating Power to Decentralise Water Management.
- **47.** Huu, Pham Cong; Rhlers, Eckart; Saravanan, V. Subramanian (2009). Dyke System Planing: Theory and Practice in Can Tho City, Vietnam.
- **48.** Evers, Hans-Dieter; Bauer, Tatjana (2009). Emerging Epistemic Landscapes: Knowledge Clusters in Ho Chi Minh City and the Mekong Delta.
- **49.** Reis, Nadine; Mollinga, Peter P. (2009). Microcredit for Rural Water Supply and Sanitation in the Mekong Delta. Policy implementation between the needs for clean water and 'beautiful latrines'.
- **50.** Gerke, Solvay; Ehlert, Judith (2009). Local Knowledge as Strategic Resource: Fishery in the Seasonal Floodplains of the Mekong Delta, Vietnam
- **51.** Schraven, Benjamin; Eguavoen, Irit; Manske, Günther (2009). Doctoral degrees for capacity development: Results from a survey among African BiGS-DR alumni.
- 52. Nguyen, Loan (2010). Legal Framework of the Water Sector in Vietnam.
- **53.** Nguyen, Loan (2010). Problems of Law Enforcement in Vietnam. The Case of Wastewater Management in Can Tho City.
- **54.** Oberkircher, Lisa et al. (2010). Rethinking Water Management in Khorezm, Uzbekistan. Concepts and Recommendations.
- **55.** Waibel, Gabi (2010). State Management in Transition: Understanding Water Resources Management in Vietnam.
- **56.** Saravanan V.S.; Mollinga, Peter P. (2010). Water Pollution and Human Health. Transdisciplinary Research on Risk Governance in a Complex Society.

- **57.** Vormoor, Klaus (2010). Water Engineering, Agricultural Development and Socio-Economic Trends in the Mekong Delta, Vietnam.
- **58.** Hornidge, Anna-Katharina; Kurfürst, Sandra (2010). Envisioning the Future, Conceptualising Public Space. Hanoi and Singapore Negotiating Spaces for Negotiation.
- **59.** Mollinga, Peter P. (2010). Transdisciplinary Method for Water Pollution and Human Health Research.
- **60.** Youkhana, Eva (2010). Gender and the development of handicraft production in rural Yucatán/Mexico.
- 61. Naz, Farhat; Saravanan V. Subramanian (2010). Water Management across Space and Time in India.
- **62.** Evers, Hans-Dieter; Nordin, Ramli, Nienkemoer, Pamela (2010). Knowledge Cluster Formation in Peninsular Malaysia: The Emergence of an Epistemic Landscape.
- **63.** Mehmood UI Hassan; Hornidge, Anna-Katharina (2010). 'Follow the Innovation' The second year of a joint experimentation and learning approach to transdisciplinary research in Uzbekistan.
- **64.** Mollinga, Peter P. (2010). Boundary concepts for interdisciplinary analysis of irrigation water management in South Asia.
- **65.** Noelle-Karimi, Christine (2006). Village Institutions in the Perception of National and International Actors in Afghanistan. (**Amu Darya Project Working Paper No. 1**)
- 66. Kuzmits, Bernd (2006). Cross-bordering Water Management in Central Asia. (Amu Darya Project Working Paper No. 2)
- **67.** Schetter, Conrad; Glassner, Rainer; Karokhail, Masood (2006). Understanding Local Violence. Security Arrangements in Kandahar, Kunduz and Paktia. (**Amu Darya Project Working Paper No. 3**)
- **68.** Shah, Usman (2007). Livelihoods in the Asqalan and Sufi-Qarayateem Canal Irrigation Systems in the Kunduz River Basin. (**Amu Darya Project Working Paper No. 4**)
- **69.** ter Steege, Bernie (2007). Infrastructure and Water Distribution in the Asqalan and Sufi-Qarayateem Canal Irrigation Systems in the Kunduz River Basin. (**Amu Darya Project Working Paper No. 5**)
- **70.** Mielke, Katja (2007). On The Concept of 'Village' in Northeastern Afghanistan. Explorations from Kunduz Province. (Amu Darya Project Working Paper No. 6)
- 71. Mielke, Katja; Glassner, Rainer; Schetter, Conrad; Yarash, Nasratullah (2007). Local Governance in Warsaj and Farkhar Districts. (Amu Darya Project Working Paper No. 7)
- 72. Meininghaus, Esther (2007). Legal Pluralism in Afghanistan. (Amu Darya Project Working Paper No. 8)
- 73. Yarash, Nasratullah; Smith, Paul; Mielke, Katja (2010). The fuel economy of mountain villages in Ishkamish and Burka (Northeast Afghanistan). Rural subsistence and urban marketing patterns. (Amu Darya Project Working Paper No. 9)
- **74.** Oberkircher, Lisa (2011). 'Stay We Will Serve You Plov!'. Puzzles and pitfalls of water research in rural Uzbekistan.
- **75.** Shtaltovna, Anastasiya; Hornidge, Anna-Katharina; Mollinga, Peter P. (2011). The Reinvention of Agricultural Service Organisations in Uzbekistan a Machine-Tractor Park in the Khorezm Region.
- **76.** Stellmacher, Till; Grote, Ulrike (2011). Forest Coffee Certification in Ethiopia: Economic Boon or Ecological Bane?
- **77.** Gatzweiler, Franz W.; Baumüller, Heike; Ladenburger, Christine; von Braun, Joachim (2011). Marginality. Addressing the roots causes of extreme poverty.
- **78.** Mielke, Katja; Schetter, Conrad; Wilde, Andreas (2011). Dimensions of Social Order: Empirical Fact, Analytical Framework and Boundary Concept.
- **79.** Yarash, Nasratullah; Mielke, Katja (2011). The Social Order of the Bazaar: Socio-economic embedding of Retail and Trade in Kunduz and Imam Sahib
- **80.** Baumüller, Heike; Ladenburger, Christine; von Braun, Joachim (2011). Innovative business approaches for the reduction of extreme poverty and marginality?
- 81. Ziai, Aram (2011). Some reflections on the concept of 'development'.
- 82. Saravanan V.S., Mollinga, Peter P. (2011). The Environment and Human Health An Agenda for Research.
- **83.** Eguavoen, Irit; Tesfai, Weyni (2011). Rebuilding livelihoods after dam-induced relocation in Koga, Blue Nile basin, Ethiopia.
- **84.** Eguavoen, I., Sisay Demeku Derib et al. (2011). Digging, damming or diverting? Small-scale irrigation in the Blue Nile basin, Ethiopia.
- **85.** Genschick, Sven (2011). Pangasius at risk Governance in farming and processing, and the role of different capital.

- **86.** Quy-Hanh Nguyen, Hans-Dieter Evers (2011). Farmers as knowledge brokers: Analysing three cases from Vietnam's Mekong Delta.
- **87.** Poos, Wolf Henrik (2011). The local governance of social security in rural Surkhondarya, Uzbekistan. Post-Soviet community, state and social order.
- **88.** Graw, Valerie; Ladenburger, Christine (2012). Mapping Marginality Hotspots. Geographical Targeting for Poverty Reduction.
- **89.** Gerke, Solvay; Evers, Hans-Dieter (2012). Looking East, looking West: Penang as a Knowledge Hub.
- **90.** Turaeva, Rano (2012). Innovation policies in Uzbekistan: Path taken by ZEFa project on innovations in the sphere of agriculture.
- **91.** Gleisberg-Gerber, Katrin (2012). Livelihoods and land management in the loba Province in south-western Burkina Faso.
- **92.** Hiemenz, Ulrich (2012). The Politics of the Fight Against Food Price Volatility Where do we stand and where are we heading?
- **93.** Baumüller, Heike (2012). Facilitating agricultural technology adoption among the poor: The role of service delivery through mobile phones.
- **94.** Akpabio, Emmanuel M.; Saravanan V.S. (2012). Water Supply and Sanitation Practices in Nigeria: Applying Local Ecological Knowledge to Understand Complexity.
- 95. Evers, Hans-Dieter; Nordin, Ramli (2012). The Symbolic Universe of Cyberjaya, Malaysia.
- **96.** Akpabio, Emmanuel M. (2012). Water Supply and Sanitation Services Sector in Nigeria: The Policy Trend and Practice Constraints.
- **97.** Boboyorov, Hafiz (2012). Masters and Networks of Knowledge Production and Transfer in the Cotton Sector of Southern Tajikistan.
- **98.** Van Assche, Kristof; Hornidge, Anna-Katharina (2012). Knowledge in rural transitions formal and informal underpinnings of land governance in Khorezm.
- **99.** Eguavoen, Irit (2012). Blessing and destruction. Climate change and trajectories of blame in Northern Ghana.
- **100.** Callo-Concha, Daniel; Gaiser, Thomas and Ewert, Frank (2012). Farming and cropping systems in the West African Sudanian Savanna. WASCAL research area: Northern Ghana, Southwest Burkina Faso and Northern Benin.
- **101.** Sow, Papa (2012). Uncertainties and conflicting environmental adaptation strategies in the region of the Pink Lake, Senegal.
- **102.** Tan, Siwei (2012). Reconsidering the Vietnamese development vision of "industrialisation and modernisation by 2020".
- 103. Ziai, Aram (2012). Postcolonial perspectives on 'development'.
- **104.** Kelboro, Girma; Stellmacher, Till (2012). Contesting the National Park theorem? Governance and land use in Nech Sar National Park, Ethiopia.
- **105.** Kotsila, Panagiota (2012). "Health is gold": Institutional structures and the realities of health access in the Mekong Delta, Vietnam.
- **106.** Mandler, Andreas (2013). Knowledge and Governance Arrangements in Agricultural Production: Negotiating Access to Arable Land in Zarafshan Valley, Tajikistan.
- **107.** Tsegai, Daniel; McBain, Florence; Tischbein, Bernhard (2013). Water, sanitation and hygiene: the missing link with agriculture.
- **108.** Pangaribowo, Evita Hanie; Gerber, Nicolas; Torero, Maximo (2013). Food and Nutrition Security Indicators: A Review.
- **109.** von Braun, Joachim; Gerber, Nicolas; Mirzabaev, Alisher; Nkonya Ephraim (2013). The Economics of Land Degradation.
- **110.** Stellmacher, Till (2013). Local forest governance in Ethiopia: Between legal pluralism and livelihood realities.
- **111.** Evers, Hans-Dieter; Purwaningrum, Farah (2013). Japanese Automobile Conglomerates in Indonesia: Knowledge Transfer within an Industrial Cluster in the Jakarta Metropolitan Area.
- **112.** Waibel, Gabi; Benedikter, Simon (2013). The formation water user groups in a nexus of central directives and local administration in the Mekong Delta, Vietnam.

- **113.** Ayaribilla Akudugu, Jonas; Laube, Wolfram (2013). Implementing Local Economic Development in Ghana: Multiple Actors and Rationalities.
- **114.** Malek, Mohammad Abdul; Hossain, Md. Amzad; Saha, Ratnajit; Gatzweiler, Franz W. (2013). Mapping marginality hotspots and agricultural potentials in Bangladesh.
- **115.** Siriwardane, Rapti; Winands, Sarah (2013). Between hope and hype: Traditional knowledge(s) held by marginal communities.
- **116.** Nguyen, Thi Phuong Loan (2013). The Legal Framework of Vietnam's Water Sector: Update 2013.
- **117.** Shtaltovna, Anastasiya (2013). Knowledge gaps and rural development in Tajikistan. Agricultural advisory services as a panacea?
- **118.** Van Assche, Kristof; Hornidge, Anna-Katharina; Shtaltovna, Anastasiya; Boboyorov, Hafiz (2013). Epistemic cultures, knowledge cultures and the transition of agricultural expertise. Rural development in Tajikistan, Uzbekistan and Georgia.
- **119.** Schädler, Manuel; Gatzweiler, Franz W. (2013). Institutional Environments for Enabling Agricultural Technology Innovations: The role of Land Rights in Ethiopia, Ghana, India and Bangladesh.
- **120.** Eguavoen, Irit; Schulz, Karsten; de Wit, Sara; Weisser, Florian; Müller-Mahn, Detlef (2013). Political dimensions of climate change adaptation. Conceptual reflections and African examples.
- **121.** Feuer, Hart Nadav; Hornidge, Anna-Katharina; Schetter, Conrad (2013). Rebuilding Knowledge. Opportunities and risks for higher education in post-conflict regions.
- **122.** Dörendahl, Esther I. (2013). Boundary work and water resources. Towards improved management and research practice?
- 123. Baumüller, Heike (2013). Mobile Technology Trends and their Potential for Agricultural Development
- **124.** Saravanan, V.S. (2013). "Blame it on the community, immunize the state and the international agencies." An assessment of water supply and sanitation programs in India.
- **125.** Ariff, Syamimi; Evers, Hans-Dieter; Ndah, Anthony Banyouko; Purwaningrum, Farah (2014). Governing Knowledge for Development: Knowledge Clusters in Brunei Darussalam and Malaysia.
- 126. Bao, Chao; Jia, Lili (2014). Residential fresh water demand in China. A panel data analysis.
- **127.** Siriwardane, Rapti (2014). War, Migration and Modernity: The Micro-politics of the Hijab in Northeastern Sri Lanka.
- **128.** Kirui, Oliver Kiptoo; Mirzabaev, Alisher (2014). Economics of Land Degradation in Eastern Africa.
- **129.** Evers, Hans-Dieter (2014). Governing Maritime Space: The South China Sea as a Mediterranean Cultural Area.
- **130.** Saravanan, V. S.; Mavalankar, D.; Kulkarni, S.; Nussbaum, S.; Weigelt, M. (2014). Metabolized-water breeding diseases in urban India: Socio-spatiality of water problems and health burden in Ahmedabad.
- **131.** Zulfiqar, Ali; Mujeri, Mustafa K.; Badrun Nessa, Ahmed (2014). Extreme Poverty and Marginality in Bangladesh: Review of Extreme Poverty Focused Innovative Programmes.
- **132.** Schwachula, Anna; Vila Seoane, Maximiliano; Hornidge, Anna-Katharina (2014). Science, technology and innovation in the context of development. An overview of concepts and corresponding policies recommended by international organizations.
- **133.** Callo-Concha, Daniel (2014). Approaches to managing disturbance and change: Resilience, vulnerability and adaptability.
- **134.** Mc Bain, Florence (2014). Health insurance and health environment: India's subsidized health insurance in a context of limited water and sanitation services.
- **135.** Mirzabaev, Alisher; Guta, Dawit; Goedecke, Jann; Gaur, Varun; Börner, Jan; Virchow, Detlef; Denich, Manfred; von Braun, Joachim (2014). Bioenergy, Food Security and Poverty Reduction: Mitigating tradeoffs and promoting synergies along the Water-Energy-Food Security Nexus.
- **136.** Iskandar, Deden Dinar; Gatzweiler, Franz (2014). An optimization model for technology adoption of marginalized smallholders: Theoretical support for matching technological and institutional innovations.
- **137.** Bühler, Dorothee; Grote, Ulrike; Hartje, Rebecca; Ker, Bopha; Lam, Do Truong; Nguyen, Loc Duc; Nguyen, Trung Thanh; Tong, Kimsun (2015). Rural Livelihood Strategies in Cambodia: Evidence from a household survey in Stung Treng.
- **138.** Amankwah, Kwadwo; Shtaltovna, Anastasiya; Kelboro, Girma; Hornidge, Anna-Katharina (2015). A Critical Review of the Follow-the-Innovation Approach: Stakeholder collaboration and agricultural innovation development.

- **139.** Wiesmann, Doris; Biesalski, Hans Konrad; von Grebmer, Klaus; Bernstein, Jill (2015). Methodological review and revision of the Global Hunger Index.
- **140.** Eguavoen, Irit; Wahren, Julia (2015). Climate change adaptation in Burkina Faso: aid dependency and obstacles to political participation. Adaptation au changement climatique au Burkina Faso: la dépendance à l'aide et les obstacles à la participation politique.
- 141. Youkhana, Eva. Postponed to 2016 (147).
- **142.** Von Braun, Joachim; Kalkuhl, Matthias (2015). International Science and Policy Interaction for Improved Food and Nutrition Security: toward an International Panel on Food and Nutrition (IPFN).
- **143.** Mohr, Anna; Beuchelt, Tina; Schneider, Rafaël; Virchow, Detlef (2015). A rights-based food security principle for biomass sustainability standards and certification systems.
- **144.** Husmann, Christine; von Braun, Joachim; Badiane, Ousmane; Akinbamijo, Yemi; Fatunbi, Oluwole Abiodun; Virchow, Detlef (2015). Tapping Potentials of Innovation for Food Security and Sustainable Agricultural Growth: An Africa-Wide Perspective.
- **145.** Laube, Wolfram (2015). Changing Aspirations, Cultural Models of Success, and Social Mobility in Northern Ghana.
- **146.** Narayanan, Sudha; Gerber, Nicolas (2016). Social Safety Nets for Food and Nutritional Security in India.
- **147.** Youkhana, Eva (2016). Migrants' religious spaces and the power of Christian Saints the Latin American Virgin of Cisne in Spain.
- **148.** Grote, Ulrike; Neubacher, Frank (2016). Rural Crime in Developing Countries: Theoretical Framework, Empirical Findings, Research Needs.
- **149.** Sharma, Rasadhika; Nguyen, Thanh Tung; Grote, Ulrike; Nguyen, Trung Thanh. Changing Livelihoods in Rural Cambodia: Evidence from panel household data in Stung Treng.
- **150.** Kavegue, Afi; Eguavoen, Irit (2016). The experience and impact of urban floods and pollution in Ebo Town, Greater Banjul Area, in The Gambia.
- 151. Mbaye, Linguère Mously; Zimmermann, Klaus F. (2016). Natural Disasters and Human Mobility.
- **152.** Gulati, Ashok; Manchanda, Stuti; Kacker, Rakesh (2016). Harvesting Solar Power in India.
- **153.** Laube, Wolfram; Awo, Martha; Derbile, Emmanuel (2017). Smallholder Integration into the Global Shea Nut Commodity Chain in Northern Ghana. Promoting poverty reduction or continuing exploitation?
- **154.** Attemene, Pauline; Eguavoen, Irit (2017). Effects of sustainability communication on environments and rural livelihoods.
- **155.** Von Braun, Joachim; Kofol, Chiara (2017). Expanding Youth Employment in the Arab Region and Africa.
- **156.** Beuchelt, Tina 2017. Buying green and social from abroad: Are biomass-focused voluntary sustainability standards useful for European public procurement?
- **157.** Bekchanov, Maksud (2017). Potentials of Waste and Wastewater Resources Recovery and Re-use (RRR) Options for Improving Water, Energy and Nutrition Security.
- **158.** Leta, Gerba; Kelboro, Girma; Stellmacher, Till; Hornidge, Anna-Katharina (2017). The agricultural extension system in Ethiopia: operational setup, challenges and opportunities.
- **159.** Ganguly, Kavery; Gulati, Ashok; von Braun, Joachim (2017). Innovations spearheading the next transformations in India's agriculture.
- **160.** Gebreselassie, Samuel; Haile Mekbib G.; Kalkuhl, Matthias (2017). The Wheat Sector in Ethiopia: Current Status and Key Challenges for Future Value Chain Development
- **161.** Jemal, Omarsherif Mohammed, Callo-Concha, Daniel (2017). Potential of Agroforestry for Food and Nutrition Security of Small-scale Farming Households.
- **162.** Berga, Helen; Ringler, Claudia; Bryan, Elizabeth; El Didi, Hagar; Elnasikh Sara (2017). Addressing Transboundary Cooperation in the Eastern Nile through the Water-Energy-Food Nexus. Insights from an E-survey and Key Informant Interviews.
- **163.** Bekchanov, Maksud (2017). Enabling Environment for Waste and Wastewater Recycling and Reuse Options in South Asia: the case of Sri Lanka.
- **164.** Kirui, Oliver Kiptoo; Kozicka, Martha (2018). Vocational Education and Training for Farmers and Other Actors in the Agri-Food Value Chain in Africa.
- **165.** Christinck, Anja; Rattunde, Fred; Kergna, Alpha; Mulinge, Wellington; Weltzien, Eva (2018). Identifying Options for the Development of Sustainable Seed Systems -Insights from Kenya and Mali.

- **166.** Tambo, Justice A. (2018). Recognizing and rewarding farmers' creativity through contests: experiences and insights from four African countries.
- **167.** Von Braun, Joachim (2018). Innovations to Overcome the Increasingly Complex Problems of Hunger.
- **168.** Maksud, Bekchanov; Evia, Pablo (2018). Sanitation system and investment climate for Resources Recovery and Reuse (RRR) options in South and Southeast Asia: The cases of India, Bangladesh, Nepal, Sri Lanka, Laos and Myanmar.
- **169.** Kirui, Oliver K.; von Braun, Joachim (2018). Mechanization in African Agriculture. A Continental Overview on Patterns and Dynamics.
- **170.** Beuchelt, Tina; Nischalke, Sarah (2018). Adding a gender lens in quantitative development research on food and non-food biomass production: A guide for sex-disaggregated data collection
- **171.** Daum, Thomas (2018). Of Bulls and Bulbs: Aspirations and perceptions of rural youth in Zambia.
- **172.** Salvatierra-Rojas, Ana; Torres-Toledo, Victor; Mrabet, Farah; Müller, Joachim (2018). Improving milk value chains through solar milk cooling.
- **173.** Desalegn, Gashaw; Ali, Seid Nuru (2018). Review of the Impact of Productive Safety Net Program (PSNP) on Rural Welfare in Ethiopia.
- **174.** Muli, Celestine; Gerber, Nicolas; Sakketa, Tekalign Gutu; Mirzabaev, Alisher (2018). Ecosystem tipping points due to variable water availability and cascading effects on food security in Sub-Saharan Africa.
- **175.** Njiraini, Georgina; Ngigi, Marther; Baraké, Evelyn (2018). Women in African Agriculture: Integrating Women into Value Chains to Build a Stronger Sector.
- 176. Bekchanov, Maksud; Evia, Pablo; Hasan, Mohammad Monirul; Adhikari, Narayan; Gondhalekar, Daphne (2018). Institutional framework and financial arrangements for supporting the adoption of Resource Recovery Reuse technologies in South Asia.
- **177.** Mirzabaev, Alisher; Njiraini, Georgina Wambui; Gebremariam, Gebrelibanos; Jourdain, Damien; Magaia, Emílio; Julio, Felita; Mosse, Gerivásia; Mutondo, João; Mungatana, Eric. Transboundary Water Resources for People and Nature: Challenges and Opportunities in the Olifants River Basin.
- **178.** Gupta, Anil; Shinde, Chintan; Dey, Anamika; Patel, Ramesh; Patel, Chetan; Kumar, Vipin; Patel, Mahesh. Honey Bee Network in Africa: Co-creating a Grassroots Innovation Ecosystem in Africa.
- **179.** Kabran, E.G. and I. Eguavoen, 2019. Ferry transportation in Abidjan. Establishment, operation and sustainability of a paratransit system.
- **180.** Sakketa, Tekalign Gutu and von Braun, Joachim. Labor-intensive public works programs in sub-Saharan Africa: Experiences and implications for employment policies.
- **181.** Engida, E., A. Srivastava, A. Kuhn and T. Gaiser, 2019. Household income implications of improved fertilizer accessibility and lower use inefficiency: Long-term scenarios for Ethiopia.
- **182.** Daum, Thomas; Capezzone, Filippo; Birner, Regina (2019). The forgotten agriculture-nutrition link: Estimating the energy requirements of different farming technologies in rural Zambia with time-use data.
- **183.** Ganguly, Kavery; Gulati, Ashok; von Braun, Joachim (2019). SkillDevelopment in the Indian Agriculture and Food Sector: A Scoping Exercise.
- **184.** Gulati, Ashok; Juneja, Ritika (2019). Agricultural Credit System in India: Evolution, Effectiveness and Innovations.
- **185.** Chaudhry, Rabia (2019). "An island of excellence"? How the Pakistan military reflects on its presence in the development sector.
- **186.** Le, Quyen Mai; Kelboro, Girma (2019). When Heritage goes ways apart: Heritagization and local involvement at the Complex of Monuments in Hue, Vietnam.
- **187.** Eguavoen, Irit; Attemene, Pauline; Kouame, Fulgence; Konan, Eugène Kouadio; Madhy, Chérif Aidara; Gleisberg-Gerber, Katrin (2019). "Dernier refuge ou presqu'île d'opportunités ? Démographie et conditions de vie à Adjahui-Coubé, une habitation spontanée à Abidjan."
- 188. von Braun, Joachim (2019). "AI and Robotics: Implications for Poverty and Marginalization."
- **189.** Daum, Thomas; Birner, Regina (2019). "African agricultural mechanizationMyths, realities and an emerging research agenda".
- **190.** Wortmann-Kolundžija, Eli (2019). "Empowering Smallholder Farmers through Farmer Organizations Insights from Kenya and Burkina Faso."

- **191.** Youkhana, Eva (2020). "Actors networks in critical urban studies protest against the subprime crisis in Madrid."
- 192. Tegegne, A. and G. Legese Feye. (2020). "Study of Selected Livestock Innovations in Ethiopia."
- **193.** Purwaningrum, F., Tayeb, A., Rahmat, S.R. and A.-K. Hornidge (2020). "Orientation shift? Understanding the 'Third Mission' of the University in Malaysia's Science System."
- **194.** Seré, C (2020). "Investing Sustainably in African Livestock Development: Opportunities and Trade-Offs."
- **195.** Gulati, A. and S. Das (2020). "India-Africa Partnership in Trade and Investment: With focus on the Agriculture and Food Sector."
- **196.** Scheiterle, L. and R. Birner (2020). "Considerations on the role of institutions and networks in the bioeconomy: three case studies from Ghana and Brazil."
- **197.** Sylla, M.B., Dimobe, K. and S. Sanfo (2021). "Burkina Faso Land, climate, energy, agriculture and development. A study in the Sudano-Sahel Initiative for Regional Development, Jobs, and Food Security."
- **198.** Admassie, A. and D. Abebaw (2021). "Ethiopia Land, climate, energy, agriculture and development. A study in the Sudano-Sahel Initiative for Regional Development, Jobs, and Food Security."
- **199.** Coulibaly, O (2021). "Mali Land, climate, energy, agriculture and development. A study in the Sudano-Sahel Initiative for Regional Development, Jobs, and Food Security.
- **200.** Adamou, R., Ibrahim, B., Bonkaney, A.L., Seyni, A.A., Idrissa, M. and N. Bellos (2021). "Niger Land, climate, energy, agriculture and development. A study in the Sudano-Sahel Initiative for Regional Development, Jobs, and Food Security."
- **201.** Olayide, O.E (2021). "Nigeria Land, climate, energy, agriculture and development. A study in the Sudano-Sahel Initiative for Regional Development, Jobs, and Food Security."
- **202.** Faye, A., Dièye, M., Diakhaté, P.B., Bèye, A., Sall, M. and M. Diop (2021). "Senegal Land, climate, energy, agriculture and development. A study in the Sudano-Sahel Initiative for Regional Development, Jobs, and Food Security.
- **203.** Osman, A. K. and A.M. Ali (2021). "Sudan Land, climate, energy, agriculture and development. A study in the Sudano-Sahel Initiative for Regional Development, Jobs, and Food Security."
- **204.** Mirzabaev A. et al (2021). "Land, Climate, Energy, Agriculture and Development in the Sahel. Synthesis paper of case studies under the Sudano-Sahelian Initiative for Regional Development, Jobs, and Food Security."
- **205.** Kampmann, W. and O.K. Kirui (2021). "Role of Farmers' Organizations in Agricultural Transformation in Africa: Overview of Continental, Regional, and Selected National Level Organizations."
- **206.** Waithaka, M, Mugoya, M., Mabaya, E. and K. Tihanyi (2021). "Decentralized Seed Services in Africa: An Assessment of Tanzania and Uganda."
- **207.** Gulati, A., Jose, S. & B.B. Singh (2021). "COVID-19: Emergence, spread and its impact on the Indian economy and migrant workers."
- **208.** Diaby-Pentzlin, F (2021). "Entwicklungspolitische Rechtsförderung, Außenwirtschaftspolitik und Gefahren für kleinbäuerliche Landwirtschaft in afrikanischen Ländern."
- **209.** von Bismarck-Osten, M (2021). "Understanding Strategic Decisions of Digital Agricultural Platform Companies. Six Case Studies of Sub-Saharan African Platforms."
- **210.** Tobón, C., Zamora, D., Ángel, M. and E. Youkhana (2021). "Deforestation in Tinigua National Natural Park: Socio-environmental consequences of the Peace Agreement in Colombia."
- **211.** Hinrichsen, E., Walakira, J.K., Langi, S., Ibrahim, N.A., Tarus, V., Badmus, O. and H. Baumüller (2022). "Prospects for Aquaculture Development in Africa: A review of past performance to assess future potential."

http://www.zef.de/workingpapers.html



Working Paper Series

Author:	Zaneta Kubik
Contact:	zkubik@uni-bonn.de
Photo:	Annie Spratt on Unsplash

Published by: Zentrum für Entwicklungsforschung (ZEF) Center for Development Research Genscherallee 3 D – 53113 Bonn Germany

Phone: +49-228-73-1861 Fax: +49-228-73-1869 E-Mail: presse.zef@uni-bonn.de

www.zef.de